Wavelet-Based Linearization Method in Nonlinear Systems

Xiaomeng Ma

A Thesis

In

The Department

of

Mechanical and Industrial Engineering

Presented in Partial Fulfillment of the Requirements

for the Degree of Master of Applied Science (Mechanical Engineering) at

Concordia University

Montreal, Quebec, Canada

July, 2010

© Xiaomeng Ma, 2010

Concordia university School of graduate studies

This is to certify that the thesis prepared

Dur	Vicomona Mo
Dy.	Alaoineng, Ma

Entitled: <u>Wavelet-based Linearization Method in Nonlinear Systems</u>

And submitted in partial fulfillment of the requirement for the degree of

Master of Applied science(mechanical engineering)

Complies with the regulations of the university and meets the accepted standards with respet to originality and quality.

Signed by the final examining committee:

Dr. AK.W,Ahmed Chair

Dr.	A.Akgunduz	Examiner

Dr. N.Bouguila Examiner

Dr. W.F.xie Supervisor

Approved by

Chair of department or graduate program director

Dean of Faculty

Date

ABSTRACT

The mathematical theories for linear systems provide a unified framework for understanding all possible linear system behaviors but no correspondingly general theoretical ones exist for treating non-linear problems. Due to the inherent nonlinearities of almost physical systems, the subject of nonlinear control is an important area of automatic control. Among many methods to deal with a nonlinear system, the most commonly used one is to linearize it which allows taking advantage of mature linear system control techniques to control the nonlinear systems approximately.

This thesis introduces a wavelet-based linearization method to estimate the nonlinear system response based on the traditional equivalent linearization technique. The mechanism by which the signal is decomposed and reconstructed using the wavelet transform is investigated. The properties and characteristics of some famous harmonic wavelet transforms are analyzed. Since the wavelet analysis can capture temporal variations in the energy and frequency content, a nonlinear system can be approximated as a time dependent linear system by combining the wavelet analysis technique with well-known traditional equivalent linearization method. The nonlinear systems including Duffing oscillator and bilinear hysteresis system are used as examples to verify the effectiveness of the proposed wavelet-based linearization method. Results from this study demonstrate that the wavelet linearization approach is more accurate and promising tool compared with traditional equivalent linearization methods in nonlinear control systems in the real world.

ACKNOWLEDGEMENTS

I would like to express my sincere gratitude to my academic supervisor and thesis adviser Professor Dr. Wen-Fang, Xie for her guidance, trust and encouragement throughout my study in Concordia University. It is impossible to complete this work without her continuous and invaluable help and support.

Heartfelt thanks go to my friends and my classmates who gave me great support in every stage at my study. In particular, I would like to thank Wang Tao, Han Xuan, and Zhao ZhongYu for their help in my study and my thesis. I would like to express my thanks to my best friend Julie for her always being around whenever I needed.

I also like to thank to the faculty at Industrial and Mechanical Engineering Department for providing great help and thoughtful services.

And the financial support that I received from Concordia University is gratefully acknowledged.

Finally, I would like to thank to my parents and my husband for everything they did for me.

TABLE OF CONTENTS

LIST OF FIGURES			
L	LIST OF 1	ГАВLES	X
I	LIST OF S	SYMBOLS, ABBREVIATIONS AND NOMENCLATURE	XI
1	INTR	ODUCTION	1
	1.1 Mo	TIVATION	1
	1.2 Овл	JECTIVES OF THE THESIS	2
	1.3 The	ESIS OUTLINE	3
2	LITE	RATURE REVIEW	5
	2.1 Tim	ie – Frequency Analysis	5
	2.1.1	Fourier Transform	5
	2.1.2	Short Time Fourier Transform	7
	2.1.3	Wigner-Ville Distribution	9
	2.1.4	Wavelets Transform	12
	2.2 Lin	EARIZATION TECHNIQUES	18
	2.2.1	Lyapunov's Linearization	19

4	2.2.2	Feedback Linearization Method	23
-	2.2.3	Describing Function Method	27
-	2.2.4	Statistical Linearization Method	32
2.3	Con	NCLUSION	34
3	WAV.	ELET TRANSFORM	35
3.1	SIG	NAL DECOMPOSITION AND RECONSTRUCTION USING THE WAVELETS	35
3.2	WAV	VELET FUNCTION DESIGN	39
-	3.2.1	Dilation Equations	40
	3.2.2	Dilation Wavelets	45
-	3.2.3	Wavelet Family Introduction	47
3.3	HA	ARMONIC WAVELET	50
3.4	Dise	CRETE HARMONIC WAVELET TRANSFORM	56
3.5	CON	NCLUSION	63
4	WAVI	ELET-BASED LINEARIZATION FOR DUFFING OSCILLATOR	
SYST	ГЕМ.		64
4.1	Dui	FFING OSCILLATOR SYSTEM MODEL	64
4.2	MU	LTI-FREQUENCY LINEARIZATION FOR STATIONARY INPUTS	68
4.3	WAV	VELET-BASED LINEARIZATION FOR DUFFING OSCILLATOR	73

	4.4	CONCLUSION	79
5	V	VAVELET-BASED LINEARIZATION FOR BILINEAR HYSTERETIC	
S	YST	EMS	80
	5.1	BILINEAR HYSTERETIC SYSTEM MODEL	80
	5.2	STATISTICAL LINEARIZATION FOR BILINEAR HYSTERETIC SYSTEM	84
	5.3	INPUT - OUTPUT RELATIONSHIP FOR A LINEAR SYSTEM	85
	5.4	WAVELET-BASED EQUIVALENT LINEARIZATION	87
	5.5	LINEARIZATION RESULTS	95
	5.6	CONCLUSION	97
6	C	ONCLUSIONS	99
	6.1	CONTRIBUTION OF THE THESIS	99
	6.2	FUTURE WORK	.100
R	EFE	RENCES	.101

LIST OF FIGURES

Figure 2. 1 Time-frequency plane partition by using STFT			
Figure 2. 2 Partition of the time-scale plane due to the wavelet transform			
Figure 2. 3 A nonlinear system			
Figure 2. 4 A linearization system			
Figure 2. 5 Detection of limit cycle			
Figure 3.1 Two-level wavelet decomposition using high-pass and low-pass filters.			
Figure 3. 2 The wavelet decomposition tree			
Figure 3. 3 Two-level wavelet reconstruction using high-pass and low-pass filters. 39			
Figure 3. 4 Construction of the scaling function			
Figure 3. 5 Construction of D4 scaling function by iteration from a box function			
over the interval x=0 to 1			
Figure 3. 6 Scaling function for D4 wavelet calculated for 12288 points using the			
method described above			
Figure 3. 7 D4 wavelet calculated by iteration			
Figure 3. 8 D4 wavelet according to (2.8) from scaling function in Figure 3.6 46			
Figure 3. 9 Haar wavelet with unit amplitude plotted for level -1, 0, 1, 2			
Figure 3. 10 Morlet and Mexican hat Wavelet forms			
Figure 3. 11 Magnitudes of the Fourier transforms of harmonic wavelets of different			
level			

Figure 3. 12 Real part of the harmonic wavelet defined by (3.14)
Figure 3. 13 Imaginary part of the harmonic wavelet defined by (3.14)
Figure 3. 14 Real part of the harmonic scaling function defined by (3.18)
Figure 3. 15 Imaginary part of the harmonic scaling function defined by (3.18) 54
Figure 3. 16 FFT algorithm to compute the harmonic wavelet transform: (a) for a
sequence of 16 real elements;[46]61
Figure 3. 17 FFT algorithm to compute the harmonic wavelet transform: (b) for a
sequence of 16 complex elements[46]
Figure 4. 1 SDOF mass-spring-damper system[54] 64
Figure 4. 2 Equivalent linearization for stationary input
Figure 4. 3 Wavelet-based linearization solution
Figure 5. 1 The bilinear system model
Figure 5. 2 The bilinear hysteresis force <i>h</i> characteristic $\alpha = \tan \varphi$
Figure 5. 3 Elasto-plastic hysteresis h_0
Figure 5. 4 The bilinear hysteresis loop
Figure 5. 5 The elaso-plastic hysteresis loop
Figure 5. 6 Numerical solution of bilinear hysteretic system to stationary excitation
Figure 5. 7 Wavelet-based linearization method comparing with numerical solution

LIST OF TABLES

 Table 2.1 Comparison with four methods in signal analysis
 17

List of Symbols, Abbreviations and Nomenclature

	Symbol	Definition
FFT		Fast Fourier Transform
IFFT		Inverse Fast Fourier Transform
STFT		Short-Time Fourier Transform
WVD		Wigner-Ville Distribution
DWT		Discrete Wavelet Transform
LTI		Linear Time Invariant
SDOF		Single Degree Of Freedom

1 INTRODUCTION

1.1 Motivation

In engineering applications, the use of a linear model for the system under consideration leads to fairly simple and often useful results when the system behaves linearly and excitation is stationary. The evaluation of the system response is not difficult. However, no real system is exactly linear. In mechanical and structural systems nonlinearities can arise in various forms and usually becomes progressively more significant as the amplitude of motion increases to excite nonlinear behavior requiring techniques which go beyond linear approach. The mathematical theories for linear systems provide a unified framework for understanding all possible linear system behaviors but no correspondingly general theoretical ones exist for treating non-linear problems. Due to the inherent nonlinearities of almost physical systems, the subject of nonlinear control is an important area of automatic control. Among many methods to deal with a nonlinear system, the most commonly used one is to linearize it which allows taking advantage of mature linear system control techniques to control the nonlinear systems approximately.

This dissertation introduces a wavelet-based method to linearize nonlinear systems to its equivalent linearization systems described by their wavelet coefficients. Time-frequency localization is the main motivation for using wavelet theory to approximate the nonlinear response as represented in this thesis.

In this dissertation, the harmonic wavelet function is chosen as the basis function in the linearization procedure due to its explicit form in mathematical expression and its non-overlapping characteristic in the frequency domain which can be interchanged from the scale to frequency band. This thesis also demonstrates that the harmonic wavelet function is used for its computational simplicity where there is a simple algorithm for harmonic wavelet analysis which uses the Fast Fourier Transform (FFT) and which is generally faster than the dilation wavelet analysis algorithm. The mechanism by which the signal is decomposed and reconstructed using the wavelet transform is investigated. The properties and characteristics of some famous harmonic wavelet transforms are analyzed.

In order to investigate the validity of this proposed method, the Duffing oscillator system and the bilinear hysteresis system are used as the examples to illustrate that the wavelet-based linearization method is a quite promising tools to find the equivalent linear parameters of the nonlinear systems in terms of time-frequency localization.

1.2 Objectives of the thesis

In this research, the main goal is to develop wavelet-based linearization method to linearize nonlinear dynamic systems. The objectives are listed as follows:

- To investigate the mechanism by which the signal is decomposed and reconstructed using the wavelet transform and to analyze the properties and characteristics of some famous harmonic wavelet transforms,
- To implement a multi-frequency linearization and wavelet-based linearization methods for Duffing oscillator system when the system is subject to multiple harmonic inputs,
- To develop a wavelet-based linearization method for the nonlinear systembilinear hysteresis system and to develop numerical algorithm to find the equivalent linear parameters.

1.3 Thesis Outline

The thesis is organized as follows. Chapter 1 introduces the motivation, objective and contributions of the thesis. The thesis outline is given as well. Chapter 2 reviews time frequency signal analysis and linearization technique as the foundation of this dissertation. Chapter 3 focuses on the wavelet transform including its mechanism by which a signal is decomposed and reconstructed using wavelet basis function, and how to obtain the wavelet function based on dilation equation method. Also some famous wavelet families are displayed as some examples to illustrate the signal decomposition at different levels and time locations. Finally, the harmonic wavelets and their significant properties are

discussed. As the most important features, these characteristics play very critical roles in the wavelet-based linearization method.

Chapter 4 gives the detailed discussion about the statistical linearization procedure for a nonlinear system including the traditional equivalent method and the multi-frequency linearization. The linearization methods are applied to Duffing oscillator system. The wavelet-based linearization approach is implemented. The effectiveness of the proposed method is verified by numerical method.

Chapter 5 addresses the wavelet-based linearization method with application in the bilinear hysteresis system and the comparison with statistical linearization method. The simulation results demonstrate that the wavelet-based linearization is a promising and applicable technique in linearization of bilinear hysteresis systems.

Chapter 6 concludes the thesis by reviewing the results. In addition, it points out the future research work.

2 LITERATURE REVIEW

In this chapter, we will review the methods related to the time-frequency analysis of signal processing. The properties and limitations of each method will be given. Also, various kinds of linearization techniques in different engineering areas will be reviewed.

2.1 Time – Frequency Analysis

The organization of this section is as follows: Section 2.1.1 starts with Fourier transform and its properties as the basic of signal analysis and then extends to Time-Frequency analysis. In Section 2.1.2, the short- time Fourier transform (STFT) and the Wigner-Ville (WV) transform are emphasized and their properties and limitations are discussed. Next, Section 2.1.3 introduces the concept of wavelets transform, its advantages and definitions as the foundation of this thesis.

2.1.1 Fourier Transform

Most engineers are familiar with the idea of frequency analysis by which a periodic function can be broken down into its harmonic components. And such a periodic function may be synthesized by adding together its harmonic components, namely, the Fourier series of function f(t) defined on the interval [-p, p] is given by:[27]

$$f(t) = a_0/2 + \sum_{n=1}^{\infty} a_n \cos \frac{n\pi}{p} t + b_n \sin \frac{n\pi}{p} t$$
(2.1)

where:

$$a_0 = \frac{1}{p} \int_p^p f(t) dt \,(1.2)$$
(2.2)

$$a_n = \frac{1}{p} \int_p^n f(t) \cos \frac{n\pi}{p} t dt$$
(2.3)

$$b_n = \frac{1}{p} \int_p^n f(t) \sin \frac{n\pi}{p} t dt$$
(2.4)

In order to obtain a representation for a non-periodic function defined for all real t, it seems desirable to take the limits as $p \to \infty$, t leads to the formulation of the famous Fourier integral theorem. Mathematically, this is continuous version of the completeness property of Fourier series. Physically, form (2.1) can be resolved into an infinite number of harmonic components with continuously varying frequency ($\frac{\omega}{2\pi}$), and amplitude $\frac{1}{2\pi}\int_{-\infty}^{\infty} f(t)e^{-i\omega\tau}d\tau$, called Fourier transform whereas the ordinary Fourier series represents a resolution of a given function into an infinite but discrete set of harmonic components.

The Fourier transform originated from the Fourier integral theorem is believed to be one of the most remarkable discoveries in the mathematical sciences and engineering applications including the analysis of stationary signals and real-time signal processing which make an effective use of the Fourier transform in time and frequency domains. The success of Fourier transform is due to the fact that under some conditions, the function f(t) can be reconstructed by the Fourier inversion formula that defines:

$$f(t) = F^{-1}\{f(\omega)\} = \frac{1}{2\pi} \int_{-\infty}^{\infty} e^{-i\omega t} f(\omega) d\omega = \frac{1}{2\pi} (f(e^{-i\omega t}))$$
(2.5)

Thus, the Fourier transform theory has been very useful for analyzing harmonic signal. On the other hand, in spite of some remarkable successes, Fourier transform analysis seems to be inadequate for studying non-stationary signal that the spectral content of the signal changes with time [51]. First, the Fourier transform of a signal does not contain any local information in the sense that reflects the change of wave number with space. Second, the Fourier transform method can be used to investigate problems either in time domain or in frequency domain, but not simultaneously in both domains. There are probably the major weaknesses of the Fourier transform analysis. It is necessary to define a single transform of time and frequency that can be used to describe the energy density of a signal simultaneously in both time and frequency domains. Such a single transform would give complete time and frequency information of signal.

2.1.2 Short Time Fourier Transform

It is well-known that the Fourier transform analysis is a very effective tool for studying stationary signal. However, signals are, in general, non-stationary, and hence cannot be analyzed completely by Fourier transform. Therefore, a complete analysis of non-stationary signal requires both time-frequency.

In order to get information about local frequency spectrum, a lot of research work have been carried out to cut the signal to segments with a short data windows centered at time t_s [70] and spectral coefficients are computed for this short length of data. The window is then moved to a new position and the calculation is repeated. This transform is called the short time Fourier transform (STFT). Based on such a procedure, Dennis Gabor [24] first introduced the windowed Fourier transform (the Gabor transform) to measure localized frequency component of signal and used a Gaussian distribution function as window function. The main idea was to use a time-localization window function $g_a(t-b)$ for extracting local information from the Fourier transform of a signal where the parameter *a* measures the width of the window and the parameter *b* is used to translate the window in order to cover the whole time domain. The property of the Gabor transform provides the local aspect of the Fourier with time resolution equal to the size of the window.[24]

$$g_{t,\omega}(\tau) = \exp(i\omega\tau)g(t-\tau)$$
(2.6)

As the window function by translating, the Gabor transform of f with the respect to g, is denoted by

$$\widetilde{f}_{g}(t,\omega) = \int_{-\infty}^{\infty} f(\tau)g(t-\tau)e^{-i\omega\tau}d\tau = \left(f,\overline{g}_{t,\omega}\right)$$
(2.7)

And the inversion formula for the Gabor transform is given by Equation (2.6). The Gabor transform $\tilde{f}_g(t, \omega)$ of a given signal f(t) depends on both time and frequency, thus the Gabor transform completely describes the spectral signal varying with time and changes an one dimensional signal into a complex value function of two real parameters: time and frequency in the two dimensional time-frequency space.

The short time Fourier Transform (STFT) remains the time information but has strong time-frequency resolution limitations. The problem with STFT can be demonstrated by Heisenberg's Uncertainty Principle [38], [71]. According to this principle, it is impossible to know the exact time -frequency representation of a signal, but one can obtain the frequency bands existing in a time interval which is a resolution problem and once the window is chosen, the frequency and time resolution are fixed for all frequencies and all times.[77]



Figure 2. 1 Time-frequency plane partition by using STFT [69]

1

2.1.3 Wigner-Ville Distribution

Wigner-Ville Distribution (WVD) is another popular tool for time-frequency analysis. It can also be thought of as a short time Fourier transform (STFT) where windowing function is time-scaled, time-reversed copy of the original signal. It is the Fourier transform of the signal's autocorrelation function with respect to the delay variable. It has much better resolution than STFT.[72]

In 1948, Ville [65] proposed the Wigner Distribution of a function or a signal f(t) in the form:

$$W_{f(t,\omega)} = \int_{-\infty}^{\infty} f\left(t + \frac{\tau}{2}\right) \bar{f}\left(t - \frac{\tau}{2}\right) e^{-i\omega\tau} d\tau$$
(2.8)

For analysis of the time-frequency structure of non-stationary signal, where $\bar{f}(t)$ is the complex conjugate of f(t), this time-frequency representation of a signal f(t) is known as the Wigner-Ville Distribution [68] which is one of the fundamental methods which have been developed over the years for the time- frequency signal analysis. In the view of its remarkable mathematical structure and properties, the Wigner-Ville Distribution is now well-recognized as an effective method for the time-frequency analysis of non-stationary signals and non-stationary random process based on the generalizing relationship between the power spectrum and autocorrelation function.

Suppose that it is possible to calculate an instantaneous correlation function:

$$R_{x}(\tau,t) = E\left[x\left(t-\frac{\tau}{2}\right)x\left(t+\frac{\tau}{2}\right)\right]$$
(2.9)

At time t, the operator E denotes the average of a statistical ensemble which is assumed

to be available for the analysis. The Fourier transform of $R_x(\tau, t)$ gives:

$$S_{x}(\omega) = \frac{1}{2\pi} \int_{-\infty}^{\infty} R_{x}(\tau, t) \exp(-i\omega\tau) d\tau$$
(2.10)

 $S_x(\omega)$ is called the spectral density of the x process and is a function of angular frequency ω . It provides information on how the mean square of the ensemble is distributed over frequency and time. But in practice, it is never possible to compute ensemble average correlation function of $R_x(\tau,t)$. Instead, the ensemble averaging operation is replaced by:

$$\phi_x(\omega,t) = \frac{1}{2\pi} \int_{-\infty}^{\infty} x \left(t - \frac{\tau}{2}\right) x^* \left(t + \frac{2}{\tau}\right) \exp\left(-i\omega\tau\right) d\tau$$
(2.11)

 $\phi_x(\omega, t)$ is called the Wigner-Ville Distribution of x and * denotes complex conjugate. From theoretical and application points of view, the Wigner-Ville Distribution [53] plays a central role and has several important structures and properties.

First, it provides a high-resolution representation in time and in frequency for some non-stationary signals. Second, it has the special property of satisfying the time and frequency analysis in terms of the instantaneous power in time and energy spectrum in frequency. In spite of these desired features, its disadvantage is that although the integral is center at time t, it covers an infinite range of τ and thus depends on the characteristic of x far away from the local t. Therefore, it does not describe the truly local behavior of x at time t, this is the fundamental uncertainty principle since time-dependent spectral high resolution cannot be obtained simultaneously in time and frequency. Also it often possesses severe cross-terms or interference terms between different time-frequency regions leading to undesirable properties.

2.1.4 Wavelets Transform

In order to overcome some of the inherent weaknesses of the short time Fourier transform and the Wigner-Ville Distribution, there has been considerable recent interest in looking for a general time-frequency distribution analysis as a mathematical method for time-frequency signal.

Wavelet Transform as the new concept can be viewed as a synthesis of various ideas originating from different disciplines including mathematic, physics and engineering [27]. The pioneer research work in wavelet transform can be found in [28], [41], [19]. Morlet (1984)[28] first introduced the idea of wavelet as a family of function constructed from translation and dilation of a single function called "mother wavelet" $\varphi(t)$ defined by:

$$\varphi_{a,b}(t) = \frac{1}{\sqrt{|a|}} \varphi\left[\frac{(t-b)}{a}\right]; \quad a, b \in \mathbb{R}, a \neq 0$$
(2.12)

where a is called a scaling parameter which measures the degree of compression or scale, and b is a translation parameter which determines the time location of the wavelet.

If |a|<1, the wavelet (2.12) is the compressed version (smaller support in time domain) of mother wavelet and corresponds mainly to higher frequencies. On the other hand, where |a|>1, $\varphi_{a,b}(t)$ has a bigger time width than $\varphi(t)$ and corresponds to lower frequencies. Thus wavelets have time-width adapted to their frequencies. This is the main reason for the success of wavelet in signal processing and time-frequency signal analysis.

It may be noted that the resolution of wavelets at different scales varies in time and frequency domains as governed by the Heisenberg uncertainty principle[71][38], that is, at large scale, the resolution is coarse in time domain and fine in the frequency domain. As the scale a decrease, the resolution in time domain becomes fine whereas the resolution in frequency domain becomes coarse.



Figure 2. 2 Partition of the time-scale plane due to the wavelet transform [69]

A more extensive study has been carried out by Grossman, [29] and the wavelet transform of function f(t) is defined by:

$$W_{\varphi}[f](a,b) = (f,\varphi_{a,b}) = \int_{-\infty}^{\infty} f(t)\overline{\varphi}_{a,b}(t)dt$$
(2.13)

where $\varphi_{a,b}(t)$ plays the same role as the kernel $\exp(i\omega t)$ in the Fourier transform. However, unlike the Fourier transform, the continuous wavelet transform is not a single transform, but any transform obtained in this way. The inverse wavelet transform can be defined so that f(t) can be reconstructed by means of the formula:

$$f(t) = \frac{1}{C_{\varphi}} \int_{-\infty-\infty}^{\infty} \int_{-\infty-\infty}^{\infty} \frac{1}{a^2} W_{\varphi}[f](a,b) \varphi\left(\frac{t-b}{a}\right) dadb$$
(2.14)

where parameter C_{φ} satisfies the so-called admissibility condition;

$$C_{\varphi} = 2\pi \int_{-\infty}^{\infty} \frac{\left|\hat{\varphi}(\omega)\right|^2}{\omega} d\omega < \infty$$
(2.15)

where parameter $\hat{\varphi}(\omega)$ is the Fourier transform of the mother wavelet $\varphi(t)$.

In practical applications involving fast numerical algorithms, the continuous wavelet can be computed at discrete grid points. To do this, a general function $\varphi_{a,b}(t)$ can be defined by replacing *a* with a_0^m ($a_0 \neq 0,1$), *b* with $nb_0a_0^m$ ($b_0 \neq 0$), where (*m*,*n*) are integers and making

$$\varphi_{m,n}(t) = a_0^{-m/2} \varphi \left(a_0^{-m} t - n b_0 \right)$$
(2.16)

The discrete wavelet transform of f(t) is defined as the

$$W_{\varphi}[f](m,n) = (f,\varphi_{m,n}) = \int_{-\infty}^{\infty} f(t)\varphi_{m,n}(t)dt \qquad (2.17)$$

If the wavelets form an orthogonal basis and complete, f(t) can be reconstructed from its discrete wavelet transform $\{f(m,n)=(f,\varphi_{m,n})\}$ by means of the formula $W_{\varphi}[f](m,n)$ $f(t) = \sum (f,\varphi_{m,n})\varphi_{m,n}(t)$ (2.18)

provided that wavelets form an orthogonal basis.

For some very special choices of φ and a_0, b_0 , the $\varphi_{m,n}$ constitute an orthogonal basis for L. In fact, if $a_0 = 2$ and $b_0 = 1$, then there exists a function φ with good time-frequency localization properties:

$$\varphi_{m,n}(t) = 2^{-m/2} \varphi(2^{-m}t - n)$$
(2.19)

It forms an orthogonal basis. These different orthogonal basis functions have been founded to be very useful in application to speech processing [73], image processing [76], computer vision and so on [66].

Generally speaking, comparing with the short time Fourier transform and the Wigner-Ville Distribution, wavelet transform has more advantages in time-frequency signal analysis due to its localization in time and frequency domains and flexible resolution to a certain extent. Meanwhile, wavelet allow complex information such as music, speech, images and patterns to be decomposed into elementary forms at different positions and scales that are generated from a single function called "mother wavelet" by

translation and dilation operations. This information is subsequently reconstructed with higher precision. Table 2.1 summarizes the comparison of four methods in signal analysis.

	Signal Analysis	Basis Function	Dimension	Resolution
Fourier Transform(FT)	stationary signal	triangle function	only one, either in the time domain or in the frequency domains	Global function without any local information in time and frequency domain.
Short time Fourier transform(STF T)	non-stationary signal	time -localization window function(a Gaussian distribution function)	two, both time and frequency domains	Local information in time and frequency but their resolutions are fixed and no any exact amount because of uncertainty principle.
Wiger-Ville Distribution(WVD)	non-stationary signal	autocorrelatio n function	two, both time and frequency domaisn	Better than STFT but similar with STFT in the uncertainty principle
Wavelet Transform(WV)	non-stationary signal	wavelet functions which are constructed from translation and dilation mother wavelet	two, both time and frequency domains	More flexible resolution and location in time and frequency due to the wavelets have time-width adapted to their frequency

Table 2.1 Comparison of Four Methods in Signal Analysis

2.2 Linearization Techniques

In many practical applications, if one wants to predict the system performance accurately, the system nonlinearity must be taken into account.

As it is mentioned previously, the importance of studying nonlinear systems in automatic control area lies in:

- No model of real system is truly linear even if the operating range of a control system is small to study its linear approximation by a linearized method.
- Nonlinear equations cannot be solved analytically and the typical analysis tools like Laplace and Fourier transform are not suitable for analyzing the nonlinear systems. As a result, nonlinear systems may demonstrate complex effects, such as limit cycles, bifurcations and even chaos which cannot be anticipated.
- In order to design an accurate and desirable controller to meet specified control performance, we need some powerful techniques to analyze such system's behaviour which must be predicted and properly compensated for.

Since the linear control is a mature subject with a plenty of powerful methods and great achievements in the industrial and engineering applications, it is natural to adopt linear control techniques to analyze nonlinear problems.

In the following sections, some typical linearization methods are introduced and, first the

Lyapunov's linearization is reviewed in Section 2.2.1. Section 2.2.2 introduces the method of feedback linearization. The describing function method is discussed in the Section 2.2.3 and the statistical linearization method is introduced in the last section.

2.2.1 Lyapunov's Linearization

The main idea of the Lyapunov's linearization technique [30] is to approximate a nonlinear system by a linear one that is around the equilibrium point, and do hope the behaviour of the solutions of the linear system will be the same as the nonlinear one.

We study Lyapunov linearization for autonomous system, and for non-autonomous systems that the parameters vary with time, we extend the research of the linearization method for non-autonomous nonlinear system.

Consider an autonomous system of second order differential equation has the form:

$$\frac{dx}{dt} = f(x, y)$$

$$\frac{dy}{dt} = g(x, y)$$
(2.20)

The constant solution to this system is called equilibrium point that satisfies:

$$f(x_0, y_0) = 0, g(x_0, y_0) = 0,$$
 (2.21)

If the system is linear with constant coefficients, we can obtain the solution of this system. However, for the nonlinear system, no method is developed for deriving a general solution to this equation.

To deal with a nonlinear problem, we must use an approximation method which is called linearization approach. As (x_0, y_0) is an equilibrium point, we try to find the linear system when (x, y) is close to (x_0, y_0) , that is, to approximate f(x, y) and g(x, y)when (x, y) is close to (x_0, y_0) . This is to calculate the functions f(x, y) and g(x, y)by the Taylor expansion approximation at a point (x_0, y_0) .

$$f(x, y) \approx f(x_0, y_0) + \frac{\partial f}{\partial x}(x_0, y_0)(x - x_0) + \frac{\partial f}{\partial y}(x_0, y_0)(y - y_0)$$
$$g(x, y) \approx g(x_0, y_0) + \frac{\partial g}{\partial x}(x_0, y_0)(x - x_0) + \frac{\partial g}{\partial y}(x_0, y_0)(y - y_0)$$
(2.22)

By replacing f(x, y) and g(x, y) in (2.20) with their linear approximations near (x_0, y_0) , we obtain:

$$\frac{dx}{dt} = f(x_0, y_0) + \frac{\partial f}{\partial x}(x_0, y_0)(x - x_0) + \frac{\partial f}{\partial y}(x_0, y_0)(y - y_0)$$
$$\frac{dy}{dt} = g(x_0, y_0) + \frac{\partial g}{\partial x}(x_0, y_0)(x - x_0) + \frac{\partial g}{\partial y}(x_0, y_0)(y - y_0)$$
(2.23)

Since (x_0, y_0) is an equilibrium point,

$$f(x_0, y_0) = g(x_0, y_0) = 0.$$

$$\frac{dx}{dt} = \frac{\partial f}{\partial x}(x_0, y_0)(x - x_0) + \frac{\partial f}{\partial y}(x_0, y_0)(y - y_0)$$

$$\frac{dy}{dt} = \frac{\partial g}{\partial x} (x_0, y_0) (x - x_0) + \frac{\partial g}{\partial y} (x_0, y_0) (y - y_0)$$
(2.24)

This is a linear system and its coefficients matrix is:

$$\frac{\frac{\partial f}{\partial x}(x_0, y_0)}{\frac{\partial g}{\partial x}(x_0, y_0)} \frac{\frac{\partial f}{\partial y}(x_0, y_0)}{\frac{\partial g}{\partial y}(x_0, y_0)}$$
(2.25)

This matrix is called Jacobian matrix of system at the point (x_0, y_0) , and the eigenvalues of the Jacobian matrix tell the information about the stability of the linear system and that of original nonlinear system.

- If the eigenvalues are negative or complex with negative real part, then the equilibrium point is stable, i. e., all solutions are convergent at the equilibrium point or if the eigenvalues are complex, then the solution will spiral around the equilibrium point. Hence, the linearized system is stable and so is the original nonlinear system.
- If the eigenvalues are positive or complex with positive real part, then the equilibrium point is unstable, i.e., all solutions will move away from the equilibrium point or if eigenvalues are complex, then the solutions will spiral away from equilibrium point. The linearized system is unstable and so is the nonlinear one.
- If the eigenvalues are real number with different sign (one positive and one

negative), then the equilibrium point is a saddle. Both of two systems are unstable.

• If the eigenvalues are complex with real part equal to zero, then the equilibrium point is center point, and all solutions form ellipses around the equilibrium point in center or at least, there is one on the imaginary axis and another is complex with negative part. The equilibrium point may be stable or unstable and hence one cannot draw any conclusion about the stability of the nonlinear system.

Note that the stability characteristics of linear systems are determined by their equilibrium point, therefore, if the nonlinear system is linearized, the local behaviour of this nonlinear system can be approximated by its linear replacing one.

The Lyapunov linearization method for non-autonomous nonlinear system involves more conditions and the powerful autonomous theorem does not apply to non-autonomous nonlinear system. However, there still exist some rules about instability for non-autonomous nonlinear system from that of its linear approximation as long as its linear approximation is time invariant.

In summary, Lyapunov's linearization method both for autonomous and non-autonomous nonlinear system provide the linearization method to analyze the nonlinear system stability issue which is critical and essential in nonlinear control system design.

2.2.2 Feedback Linearization Method

Feedback linearization is a common approach [26] used in nonlinear control systems. The idea of this method is to transform original system model into equivalent linear one which is changed to the state variables and a suitable input instead.

This method is different from the Jacobian linearization mentioned previously, which is approximated by its linearized system, actually, feedback linearization method transforms the exact state variables in some algebraic forms so that the nonlinear system is changed into a linear one and linear control techniques can be applied.

As one of the most important control techniques, the feedback linearization is applied successfully in the control of high performance aircraft, industrial robots and so on.

This part provides a procedure of feedback linearization including the basic concepts of this method and mathematical tools to transform nonlinear components into linear systems.

First, the basic concept of feedback linearization is to cancel the nonlinearities in a nonlinear system so that the closed-loop dynamics is in a linear form [37].

The nonlinear system has the forms:

$$x^{(n)} = f(X) + b(X)u$$
(2.26)

where *u* is the scalar control input, *x* is the scalar output, and $X = [x, \dot{x}, ..., x^{(n-1)}]^T$ is the state vector, and f(X) and b(X) are nonlinear function of the states. The Equation (2.26) can be written as the form:

$$\frac{d}{dt} \begin{bmatrix} x_1 \\ \dots \\ x_{n-1} \\ x_n \end{bmatrix} = \begin{bmatrix} x_2 \\ \dots \\ x_n \\ f(X) + b(X)u \end{bmatrix}$$
(2.27)

Thus the system is expressed in the controllability canonical form, and the input u is designed as:

$$u = \frac{1}{b} [v - f] \quad b \neq 0,$$
(2.28)

We cancel the nonlinearities and obtain the simple input –output relationship in the form that is:

$$x^{(n)} = v$$
, (2.29)

We can design the control law that is:

$$v = -c_0 x - c_1 \dot{x} - \dots - c_{n-1} x^{(n-1)}$$
(2.30)

With the parameters c_i chosen so that the polynomial $p^n + c_{n-1}p^{n-1} + ... + c_0$ has all its roots strictly in the left-half complex plane, the nonlinear system leads to the exponentially stable dynamics.

$$x^{(n)} + c_{n-1}x^{(n-1)} + \dots + c_0x = 0$$
(2.31)

If the control of the tracking problem is considered, the control law can be:

$$v = x_d^{(n)} - c_0 e - c_1 \dot{e} - \dots - c_{n-1} e^{(n-1)}$$
(2.32)

where $e(t) = x(t) - x_d(t), x_d(t)$ is a desired output.

Second, some mathematical tool related to feedback linearization technique is introduced. Lie derivatives:

Consider nonlinear system:

$$\dot{x} = f(x) + g(x)u$$

$$y = h(x)$$
(2.33)

where $x \in \mathbb{R}^n$ is the state vector, $u \in \mathbb{R}^p$ is the vector of inputs, and $y \in \mathbb{R}^m$ is the

vector of outputs.

$$\dot{y} = \frac{dh(x)}{dx} \cdot \frac{dx}{dt} = \frac{dh(x)}{dx} \cdot \dot{x} = \frac{dh(x)}{dx} f(x) + \frac{dh(x)}{dx} g(x)u$$
(2.34)

Defining the lie derivative h(x) along f(x) as

$$L_f h(x) = \frac{dh(x)}{dx} f(x), \qquad (2.35)$$

And similarly, the lie derivative h(x) along g(x) as

$$L_g h(x) = \frac{dh(x)}{dx} g(x)$$
(2.36)

Thus,
$$\dot{y} = L_f h(x) + L_g h(x) u$$
 (2.37)

The goal of the feedback linearization is to find a direct and simple relationship between the system output y and the control input u by differentiating the output y until the input u
is expressed in the same equation. Thus we obtain:

$$z = T(x) = \begin{bmatrix} z_1(x) \\ z_2(x) \\ ... \\ z_n(x) \end{bmatrix} = \begin{bmatrix} y \\ \dot{y} \\ ... \\ y^{(n-1)} \end{bmatrix} = \begin{bmatrix} h(x) \\ L_f h(x) \\ ... \\ L_f^{n-1} h(x) \end{bmatrix}$$
(2.38)

$$\begin{cases} \dot{z}_{1} = L_{f}h(x) = z_{2}(x) \\ \dot{z}_{2} = L_{f}^{2}h(x) = z_{3}(x) \\ \dots \\ \dot{z}_{n} = L_{f}^{n}h(x) + L_{g}L_{f}^{n-1}h(x)u \end{cases}$$
(2.39)

We assumed the relative degree of system is *n*, the lie derivative of the form $L_g L_f^i h(x)$ for *i*=1...*n*-2 are all zeros. And the original X coordinate is transformed into the Z coordinate so that the new linearized system with z variables is obtained as well as the relationship between input and output. Therefore, the control law is that:

$$u = \frac{1}{L_g L_f^{n-1} h(x)} \Big(L_f^n h(x) + v \Big)$$
(2.40)

Also the input-output relationship from v to $z_1 = y$ is that

$$\begin{vmatrix} \dot{z}_1 = z_2 \\ \dot{z}_2 = z_3 \\ \cdots \\ \dot{z}_n = v \end{vmatrix}$$
(2.41)

Choosing the suitable C in v = -Cz with standard linear system methodology, we get the linear system with the linearizing state z and linearizing control law v. In summary, feedback linearization method is based on the idea of transforming nonlinear system into a linear one by using state feedback with input-state linearization or input-output linearization method which can be used for both stabilization and tracking control problems and has been applied in many practical nonlinear controlling systems to achieve successful results. However, this linearization method needs the exact nonlinear function in the system dynamics which is usually difficult to obtain in the real practice. In addition, the controller design based on the linearized model will cause infinity control action if $L_e L_p^n {}^1h(x)$ in (2.40) becomes very small.

2.2.3 Describing Function Method

Basically, the describing function method is the statistical extension of [40] linearization technique in the electrical engineering literature which is an approximation procedure for analyzing certain nonlinear control problems that were originally developed by [12] and used as a tool in control engineering. Subsequent developments of this method in this field have described in [56] and [25].

The describing function method is an approximation procedure for analyzing certain nonlinear control problems. It is based on replacing each nonlinear element with a quasi-linear descriptor which is the approximation of nonlinear system by a linear transfer function whose gain is determined by the input amplitude while a true linear system transfer function does not depend on that amplitude. And describing function can also be treated as an extended version of the frequency response method to predict nonlinear behavior, especially, for limit cycle prediction in nonlinear systems. The describing function is a powerful tool to discover the existence of limit cycles and determine their stability. [62]

Before discussing the property of describing function, some assumptions must be given to satisfy some conditions:

- There is only a single nonlinear component.
- The nonlinear component is the time invariant.
- Corresponding to a sinusoidal input $x = sin(\omega t)$, only the fundamental component $y_1(t)$ in the output y(t) has been considered. This assumption implies that higher frequency harmonics can be neglected, and the approximated linear one has low-pass properties.
- The nonlinearity is odd that means the relationship with input and output of nonlinear element is symmetric about origin so that the static term in the Fourier expansion of output can be ignored.

A. Basic Definitions

Consider a sinusoidal input $x = A\sin(\omega t)$ to nonlinear element.

where A is amplitude and ω is frequency, as show in Figure 2.3.



Figure 2. 3 A nonlinear system

The output y(t) is always a periodic function that is extended by Fourier series expressed as:

$$y(t) = \frac{a_0}{2} + \sum_{n=1}^{\infty} [a_n \cos(n\omega t) + b_n \sin(n\omega t)]$$
(2.42)

where the Fourier coefficients a_n, b_n are determined by

$$a_{0} = \frac{1}{\pi} \int_{-\pi}^{\pi} y(t) d(\omega t)$$
 (2.43)

$$a_n = \frac{1}{\pi} \int_{-\pi}^{\pi} y(t) \cos(n\omega t) d(\omega t)$$
(2.44)

$$b_n = \frac{1}{\pi} \int_{-\pi}^{\pi} y(t) \sin(n\omega t) d(\omega t)$$
(2.45)

Due to the fourth assumption implying that $a_0 = 0$, the third assumption implies that only the fundamental component $y_1(t)$ is considered:

$$y(t) \approx y_1(t) = a_1 \cos(\omega t) + b_1 \sin(\omega t) = M \sin(\omega t + \phi)$$
(2.46)

where
$$M(A, \omega) = \sqrt{a_1^2 + b_1^2}$$
 and $\phi(A, \omega) = \arctan\left(\frac{a_1}{b_1}\right)$.

And also this output can be written as :

$$y_1(t) = Me^{j(\omega t + \phi)} = (b_1 + ja_1)e^{j(\omega t + \phi)}$$

Since the concept of frequency response function is the frequency ratio of the sinusoidal input and the sinusoidal output of system, we define the describing function of nonlinear element to be complex ratio of the fundamental component of nonlinear element by input sinusoidal, that is:

$$N(A,\omega) = \frac{Me^{j(\omega t+\phi)}}{Ae^{j\omega t}} = \frac{M}{A}e^{j\phi} = \frac{1}{A}(b_1 + ja_1)$$
(2.47)

Thus, the describing function [64] can be treated as the approximation of linear one with a frequency response function $N(A, \omega)$, which is considered as an extension of the concept of frequency response function. However, the describing function of nonlinear element differs from linear system frequency response function in that it depends on the input amplitude A.

B. Describing Function Analysis Nonlinear System

As we know how to get the describing function for the nonlinear element, we use this tool to predict the limit cycles that is based on the linear approach of the famous Nyquist criterion in control system to investigate the stability.

Consider a self-sustained oscillation of amplitude A and frequency ω in the system of

Figure 2.4.



Figure 2. 4 A linearization system

The variables in the loop must satisfy the following relations:

$$x = y$$

$$w = N(A, \omega)x$$

$$y = G(j\omega)w$$

Therefore, $y = G(j\omega)N(A, \omega)(-y)$. And $y \neq 0$. (2.48)

So, $G(j\omega)N(A,\omega)+1=0$ and it can be written as:

$$G(j\omega) = -\frac{1}{N(A,\omega)}$$
(2.49)

The amplitude A and frequency ω of the limit cycles must satisfy (2.49), if the above equation has no solution and the nonlinear system has no limit cycles.

However, it is difficult to get the solution by this analytical approach, especially, for higher order system, thus the graphical way is usually used. The idea is to plot both sides of (2.49) in complex plane and find the intersection points of two curves. And also according to Nyquist criterion, we can determine the stability of limit cycles as shown in Figure 2.5.



Figure 2. 5 Detection of limit cycle

Generally speaking, the describing function method which is an extension of the frequency response method of linear control can be used as the approximate analysis to predict some important characteristics of nonlinear system including systems with hard nonlinearities such as, saturation, backlash and hysteresis [1]. Particularly, it is the main tool to predict the limit cycles in the nonlinear system in the graphical nature of frequency domain [52].

2.2.4 Statistical Linearization Method

Among those various possible approaches which are available, the statistical method or equivalent linearization method has proven to be very useful approximate technique in structure dynamics and earthquake engineering [6].

Methods of predicting the random vibration response of mechanical and structural system

to random external forces has grown rapidly. Since the excitation and response of such system are always non-periodic, highly irregular and no repeatability, linear method is not suitable to apply, and new techniques are required.

However, in engineering application, because the linear system is fairly simple and there a lot of resources to obtain the useful and desired solution, the systems can be modeled in terms of the linear differential equations of motion. Although no real system is exactly linear, in mechanical and structural systems, non-linearity can arise in various forms and usually become more significant as the amplitude of vibration increase. In order to predict the response of this kind of system or to get an approximation solution of non-linear equation, the statistical method or equivalent linearization method is applied to estimate the accurate equivalent linear parameters.

The statistical equivalent linearization method is based on the idea that the non-linear system is replaced by an equivalent linear equation by minimizing the difference between the two systems in some appropriate sense. This technique has been used for deterministic non-linear problems for many years, for instance, referred to as "describing function" method [40], and the adaptation of approach to deal with stochastic problem was first developed by [12] to use as a tool in control engineering And subsequent developments of this field have been described by[56],[25],[57]. This method known as

statistical linearization or equivalent linearization was proposed by [14], [15]as a way to solve nonlinear vibration problems, furthermore, the statistical method is applied and served as the fundamental concept[58],[59] to propose the novel wavelet-based linearization method.[8],[9]

In summary, this part reviews some typical linearization methods and their applications in the nonlinear system control design and analysis. Comparing with their properties and limits, it provides fundamental background knowledge for the proposed linearization method that is based on the statistical linearization approach. The details about this method will be discussed in the Chapter 4 & 5 by using some typical nonlinear systems as the examples.

2.3 Conclusion

In this chapter, comprehensive review and introduction on time-frequency analysis of signal processing and linearization techniques are given. The properties and limitations of each method have been analyzed.

3 WAVELET TRANSFORM

This chapter presents the mechanism by which the signal is decomposed and reconstructed using the wavelet transform and introduces some famous wavelet basis functions. The properties and characteristics of the harmonic wavelet transform are discussed. The discrete harmonic wavelet transform will be also described.

3.1 Signal Decomposition and Reconstruction Using the Wavelets

The discrete wavelet scheme is used for this presentation due to its simplicity and ease of implementation. In discrete wavelet transform (DWT), a time-scale representation of a digital signal is obtained using digital filtering techniques [44]. At each of filtering process, a signal is decomposed in high scale-low frequency components which are called approximations (A) and in low scale-high frequency components which are called details (D). This signal is passed through a series of low- pass filters to analyze the low frequencies and pass through a series of high- pass filters to analyze the high frequencies. During the decomposition process, twice the amount of the original signal data is generated at each stage. To avoid this result, the data has to been down-sampled by a factor of 2 that is shown as:

$$f(t) - g(t) = f(2t)$$

That means a discrete time signal f(t) consists of omitting every other value and we can think of a system whose input is f(t) and whose output is g(t)=f(2t). The same calculation holds for the detail coefficients.

The filter process is demonstrated in Figure 3.1



Figure 3.1 Two-level wavelet decomposition using high-pass and low-pass filters.

The decomposition process can be iterated with successive approximations being decomposed in turn [42] so that one signal is broken down into many lower resolution components. This is called wavelet decomposition tree which is shown in Figure 3.2



Figure 3. 2 The wavelet decomposition tree

The decomposition lasts until the detail coefficients consist of repetitions of a single sample. To synthesize the signal, the inverse process of up sampling and filtering is followed by using the low (L') and high (H') pass reconstruction filters. Up sampling of discrete time signal f(t) consists of inserting zeros between the values. We can think about a system with input f(t) and output g(t)=f(t/2) for even values of t, and g(t)=0 for odd values of t shown as:

$$f(n) \rightarrow \underbrace{\uparrow 2}_{\text{up sampled}} \rightarrow g(n) = \begin{bmatrix} f(\frac{n}{2}) \text{ for } n \text{ even} \\ 0 \text{ for } n \text{ odd} \end{bmatrix}$$

The design of the low and high pass filter for the decomposition and reconstruction process is very important because it determines the shape of the wavelet function used in analysis.[61],[20],[44],[74],[75] In fact, the filter design and wavelet function choice

develop at the same time.

The filters used in the decomposition and reconstruction process are different in that they must be satisfied some conditions:[20]

Condition 1:

$$H\left(\omega + \frac{1}{2}\right)H'(\omega) + L\left(\omega + \frac{1}{2}\right)L'(\omega) = 0$$
(3.1)

where $H(\omega)$ and $L(\omega)$ are Fourier transform of high-pass filter *h* and low-pass filter *l*, also H' and L' are conjugate to H and L and the same as h and l to h' and l'. It is defined by,

$$H(\omega) = \sum_{n} h(n) e^{-2\pi n\omega}$$
(3.2)

And

$$L(\omega) = \sum_{n} g(n) e^{-2\pi n\omega}$$
(3.3)

Condition 2:

$$H(\omega)H'(\omega) + L(\omega)L'(\omega) = 1$$
(3.4)

In the signal decomposed or reconstructed process, the function is involved into successive two filters L (low frequency) and H(high frequency) or their conjugate filter L' and H' that so-called quadric mirror filters.



Figure 3. 3 Two-level wavelet reconstruction using high-pass and low-pass filters.

There are infinite numbers of possibility to choose the analyzing wavelet. However, only a small group of these meets the conditions that are necessary if the wavelets are given an accurate decomposition and reconstruction and also be orthogonal to each other[45]. This is extremely critical and efficient for computing wavelet transform which does for wavelet analysis the same as what FFT (fast Fourier transform) does for Fourier analysis.

3.2 Wavelet Function Design

As mentioned previously, the wavelet function design is based on accuracy and calculating efficiency. There are two classes of orthogonal wavelets-- Dilation wavelet and harmonic wavelet.

First we consider the theory of dilation wavelets and later discuss harmonic wavelet.

The successful development of dilation wavelet in mathematical community has been led

by Daubechies[19], [20], [21] who invented so-called dilation schemes and for first time gave the method to calculate the dilation wavelet coefficients.

3.2.1 Dilation Equations

To dilate is to spread out so that dilation means expansion [61]. The basis function φ (x) is a dilated horizontally version φ (2x) that has the same height but is stretched over twice the horizontal scale of x, where x is a independent variable that may represent time or position depending on the applications. In dilation equation, φ (x) is expressed as a finite series of terms, each of which involves φ (2x). Each of these φ (2x) terms is positioned at different places on the horizontal axis by making the translation (2x-k) instead of 2x, where k is an integer(positive or negative).

The dilation equation has a form:

$$\phi(x) = c_0 \phi(2x) + c_1 \phi(2x-1) + c_2 \phi(2x-2) + c_3 \phi(2x-3)$$
(3.5)

where c's are numerical constants. It is impossible to solve (3.5) directly to find function $\varphi(x)$ except for very simple case. But indirectly setting up an iterative algorithm in which each new approximation $\phi_j(x)$ is calculated from the previous approximation $\phi_{j-1}(x)$ by the scheme:

$$\phi_j(x) = c_0 \phi_{j-1}(2x) + c_1 \phi_{j-1}(2x-1) + c_2 \phi_{j-1}(2x-2) + c_3 \phi_{j-1}(2x-3)$$
(3.6)

The iterative process continues until $\phi_j(x)$ becomes the same as $\phi_{j-1}(x)$.

Consider starting from a box function:

$$\phi_0 = 1$$
 $0 \le x < 1;$

$$\phi_0 = 0$$
, elsewhere.

After one iteration the box function over interval x=0 to 1 has expanded into a staircase function over the interval x=0 to 2 also shown in the Figure 3.4

			Ø ₀ (x)	
			C	₀ Ø₀ (2x)	
			С	₁ ∅₀ (2x-1)	
			C	₂Ø₀ (2x-2)	
			C	₃ Ø₀ (2x-3)	
			Ø	1 (X)	
 0.5	1	1.5	2	2.5	;
		Y			

Constrution of \emptyset_1 (x) = $C_0 \emptyset_0(2x) + C_1 \emptyset_0(2x-1) + C_2 \emptyset_0(2x-2) + C_3 \emptyset_0(2x-3)$

Figure 3. 4 Construction of the scaling function

Each contribution to $\phi_1(x)$ is shown separately and then four contributions are shown added together. A particular set of the coefficients $c_{0,}c_1,c_2,c_3$ have been obtained as follows:

$$c_{0} = (1 + \sqrt{3})/4; c_{1} = (3 + \sqrt{3})/4;$$

$$c_{2} = (3 - \sqrt{3})/4; c_{3} = (\sqrt{3} - 1)/4;$$
(3.7)

This is the orthogonal D4 wavelet. The D stands for Daubechies.

The function $\phi(x)$ coming from a unit box (unit height and length) is called scaling function and a corresponding wavelet function will be obtained from it. When the iterative process continues, it has been seen that the first iteration produces the four coefficients $c_{0,c_{1}}, c_{2}, c_{3}$ at x=0, 0.5, 1, 1.5. At the second process, each coefficient produces another four coefficients. For example, c_{0} at x=0 produces four new coefficients which are $c_{0}^{2}, c_{0}c_{1}, c_{0}c_{2}, c_{0}c_{3}$ at x=0, 0.25, 0.5, 0.75; and c_{1} at x=0.5 contributes to four new coefficients $c_{1}c_{0}, c_{1}^{2}, c_{1}c_{2}, c_{1}c_{3}$ at x=0.5, 0.75, 1, 1.25; and so on. After second iteration, all the resulting ordinates corresponding to the coefficients are obtained. Figure 3.5 displays the first eight iterations of the development of the D4 scaling function. Figure 3.6 shows scaling function for D4 wavelet calculated for 12288 points.



Figure 3. 5 Construction of D4 scaling function by iteration from a box function over the

interval x=0 to 1

From this process we find the matrix scheme

$$\begin{bmatrix} \phi_{1} & & & & \\ c_{1} & & & & \\ c_{2} & c_{0} & & & \\ c_{3} & c_{1} & & & \\ & c_{2} & c_{0} & & \\ & c_{3} & c_{1} & & \\ & & c_{3} & c_{1} & & \\ & & & c_{3} & & \\ & & & c_{3} & & \\ & & & & c_{3} & & \\ & & & & & c_{3} & & \\ & & & & & & \\ \end{bmatrix} \begin{pmatrix} c_{0} & & \\ c_{1} & & \\ c_{2} & & \\ c_{2} & & \\ c_{3} & & \\ c_$$

where M_r denotes a matrix of order $(2^{r+1}+2^r-2)\times(2^r+2^{r-1}-2)$ in which each column has a sub-matrix of coefficients $c_{0,}c_1,c_2,c_3$ positioned two places below the sub-matrix to left.

The number of points increases in the sequence and after eight iterations it reaches 766 points and each of them spaced $1/2^8 = 1/256$. Although the iteration method is not an efficient way, it is simple to understand and easily to be programmed.



Figure 3. 6 Scaling function for D4 wavelet calculated for 12288 points

3.2.2 Dilation Wavelets

The wavelet function W(x) is derived from scaling function by taking differences. For the four –coefficient scaling function defined by (3.5) the dilation wavelet function is $w(x) = -c_3\phi(2x) + c_2\phi(2x-1) - c_1\phi(2x-2) + c_0\phi(2x-3) \qquad (3.8)$

The same coefficients are used as scaling function $\phi(x)$ but in reverse order and with alternate terms having their signs changing from plus to minus.

The results of making the calculation (3.8) for D4 scaling function in Figure 3.6 are shown in Figure 3.8 and the iteration of development of the D4 wavelet is shown in Figure 3.7.

So far using the scaling function $\phi(x)$ to construct orthogonal wavelet function which has been developed by Daubeches[19] [20], [21] who first computed the values of required number N of coefficients $c_0, c_1, ..., c_{N-1}$ in order to achieve accurate level expansion for signal decomposition and reconstruction







Figure 3. 8 D4 wavelet according to (2.8) from scaling function in Figure 3.6

3.2.3 Wavelet Family Introduction

There is a lot of wavelet functions which can be used as the mother wavelet function to achieve a signal's wavelet transform. Among those choices we introduce some very commonly used wavelets in engineering applications. [38]

First, it is the Haar wavelet which is the simplest orthogonal basis function. Figure 3.9 shows how a signal is decomposed in levels.



Figure 3. 9 Haar wavelet with unit amplitude plotted for level -1, 0, 1, 2

The Haar wavelet function is:
$$\begin{cases} w(x) = -1 \text{ for } 0 \le x \le 1/2 \\ w(x) = 1 \text{ for } 1/2 \le x \le 1 \end{cases}$$
(3.9)

The time or position x is plotted along the horizontal axis and wavelet function w(x) is 0 to 1 which are located at different position along x axis.

There are two half length wavelets represented by w(2x) and w(2x-1) and four-quarter length wavelets represented by w(4x), w(4x-1), w(4x-2), w(4x-3), as shown in Figure 3.9. The scale and position can be obtained from its argument. For example, w(2x-1) is the same as w(x) except that it is compressed into half the horizontal length and it begins at x=1/2 instead of x=0.

The wavelet's level is determined by how many wavelets fit into the unit interval x=0 to1. At level 0 there is $2^0 = 1$ wavelet like the third top view of Figure 3.9. And there are $2^1 = 2$ wavelets which fit between 0 to 1 and at level 2, there are $2^2 = 4$ wavelets which fit between 0 to 1 like the top view Figure 3.9 and so on.

The decomposition of a signal or function f(x) is the same as its Fourier transform in which the sequence length N of signal being analyzed determines how many separate frequencies can be represented. In the wavelet transform, the sequence length determines how many wavelet levels there are with different scales and positions in which the sequence of N=2^{*n*} points consist of n+1 levels (scales) running from -1 to n-1 and 2^{*n*-1} position.

$$f(x) = a_0 + a_1 w(x) + a_2 w(2x) + a_3 w(2x-1) + a_4 w(4x) + a_5 w(4x-1) + a_6 w(4x-2) + a_7 w(4x-3) + \dots$$
(3.10)
$$= a_0 + \sum_{j=0}^{\infty} \sum_{k=0}^{2^j - 1} a_{2^j + k} w(2^j x - k)$$

There are some other wavelet family functions such as Mexcian hat wavelet whose analytical expression is:[21]

$$\varphi(x) = \left(\frac{2}{\sqrt{3}}\pi^{\frac{1}{4}}\right) (1 - x^2) e^{-x^2/2}$$
(3.11)

And Morlet wavelet is defined as:

$$\varphi(x) = e^{-x^2/2} e^{j\omega_0 t}$$
 where $\omega_0 = 1.75\pi$ (3.12)

Both basis wavelet functions shown in Figure 3.10 are applied in different signals analysis and other engineering applications. Therefore mother wavelets choice should take into account the details and particular application so that the wavelet transform can represent function characteristics effectively.



Figure 3. 10 Morlet and Mexican hat Wavelet forms

3.3 Harmonic Wavelet

In this section harmonic wavelet is introduced and its properties and characteristics are discussed as the fundamental concept to construct the rest of this thesis. As the number N of wavelet coefficients is increased, the wavelet's Fourier transform becomes compact. Only the values of N up to 20 are included in the program. However the number of wavelet coefficients for N> 20 is required to be computed and it turns out that the spectrum of a wavelet with N coefficients are more like box-like as N increases. This fact lead Newland [46] to seek a wavelet w(x) whose spectrum is exactly like a box so that the magnitude of Fourier transform $w(\omega)$ is zero except for an octave band of frequencies. A simple way to define |w(x)| is shown in Figure 3.11.



Figure 3. 11 Magnitudes of the Fourier transforms of harmonic wavelets of different level

All the Fourier transforms are identically zero for $\omega < 0$, and for level zero the definition of |w(x)| is: $|w(\omega)| = \frac{1}{2\pi} \text{ for } 2\pi \le \omega < 4\pi$ = 0elsewhere (3.13)

By calculating the inverse Fourier transform of $w(\omega)$, we obtain the corresponding complex wavelet as:

$$w(x) = \frac{\left(e^{i4\pi x} - e^{i2\pi x}\right)}{i2\pi x}$$
(3.14)

whose real and imaginary parts are shown in Figure 3.12 and Figure 3.13



Figure 3. 12 Real part of the harmonic wavelet (3.14)



Figure 3. 13 Imaginary part of the harmonic wavelet (3.14)

For the general complex wavelet, at level j and translated by k step of size $\frac{1}{2^{j}}$ we define:

$$w(\omega) = \left(\frac{1}{2\pi}\right) 2^{-j} e^{\frac{i\omega k}{2^{j}}} \text{ for } 2\pi 2^{j} \le \omega < 4\pi 2^{j}$$

= 0elsewhere (3.15)

By calculating the inverse Fourier transform, this gives:

$$w(2^{j}x-k) = \frac{\left(e^{i4\pi(2^{j}x-k)} - e^{i2\pi(2^{j}x-k)}\right)}{i2\pi(2^{j}x-k)}$$
(3.16)

where j = 0 to $\infty k = \infty$ to ∞ . For level -1 with the frequency band $0 \le \omega < 2\pi$ in Figure 3.11 the wavelet is defined as

$$w(\omega) = \binom{1}{2\pi} e^{-i\omega k} \text{ for } 0 \le \omega < 2\pi$$

= 0elsewhere (3. 17)

The scaling function is defined as:

$$\phi(x) = \frac{\left(e^{i2\pi(x-k)} - 1\right)}{i2\pi(x-k)}$$
(3.18)

 $k = \infty to \infty$, the real and imaginary parts of $\phi(x)$ are shown in Figure 3.14 and Figure





Figure 3. 14 Real part of the harmonic scaling function (3.18)



Figure 3. 15 Imaginary part of the harmonic scaling function (3.18)

The idea to choose the wavelet and scaling function is that they form the orthogonal set and have the properties:

$$\int_{-\infty}^{\infty} w (2^{j} x - k) w^{*} (2^{r} x - s) dx = 0 \quad \text{for all } j, k, s; j, r \ge 0$$
(3.19)

$$\int_{-\infty}^{\infty} w (2^{j} x - k) w (2^{r} x - s) dx = 0 \quad \text{for all } j, k, r, s; j, r \ge 0$$
(3.20)

Except for the case when r=j, and s=k. when $\int_{-\infty}^{\infty} |w(2^{j}x-k)|^{2} dx = \frac{1}{2^{j}}$

Also
$$\int_{-\infty}^{\infty} \phi(x-k)\phi(x-s)dx = 0 \quad \text{for all } k, s \tag{3.21}$$

$$\int_{-\infty}^{\infty} \phi(x-k)\phi^*(x-s)dx = 0 \quad \text{for all } k, s \tag{3.22}$$

Except s=k, when $\int_{-\infty}^{\infty} |\phi(x-k)|^2 dx = 1$ And finally, $\int_{-\infty}^{\infty} w(2^j x - k)\phi(x-s)dx = 0$ for all j, k, s; (j ≥ 0) (3.23)

$$\int_{-\infty}^{\infty} w(2^{j} x - k) \phi^{*}(x - s) dx = 0 \text{ for all } j, k, s; (j \ge 0)$$
(3.24)

The proof of these results depends on the properties of their Fourier transform and derives from two results from Fourier transform theory of w(x) and v(x) respectively, then one has [48]

$$\int_{-\infty}^{\infty} w(x)v(x)dx = 2\pi \int_{-\infty}^{\infty} W(\omega)V(-\omega)d\omega$$
(3.25)

$$\int_{-\infty}^{\infty} w(x) v^*(x) dx = 2\pi \int_{-\infty}^{\infty} W(\omega) V^*(\omega) d\omega$$
(3.26)

From Equations(3.15),(3.16),(3.17), the wavelets of different of levels have Fourier transform that occupy different frequency band while for the wavelets in the same frequency band, each wavelet is only orthogonal to the complex conjugate of another wavelet.

The result of all this is that the $w(2^{j}x-k)$ defined by (3.16) together with $\phi(x-k)$ defined by (3.18) construct an orthogonal family of wavelet that offer an alternative to wavelet family from dilation equation.

The differences of two wavelet functions are that Harmonic wavelet can be described by a simple analytical formula which are compact in frequency domain and can be described by a complex function so that there are two real wavelets for each j, k pair. While Dilation wavelets can not be expressed in functional form. They are compact in x-domain and there is one real wavelet for each j, k pair.

Harmonic wavelet has an advantage that is computation simplicity which uses FFT (fast

Fourier transform) to harmonic wavelet analysis to a signal or function.

3.4 Discrete Harmonic Wavelet Transform

Because of complex wavelets, two amplitude coefficients have to be defined as follows:[46]

$$a_{j,k} = 2^{j} \int_{-\infty}^{\infty} f(x) w^{*} (2^{j} x - k) dx$$
(3.27)

$$\widetilde{a}_{j,k} = 2^j \int_{-\infty}^{\infty} f(x) w \left(2^j x - k \right) dx$$
(3.28)

If f(x) is a real function, $\tilde{a}_{j,k}$ is the complex conjugate of $a_{j,k}$. However, if f(x) is complex $\tilde{a}_{j,k}$ and $a_{j,k}$ are different.

Similarly,

$$a_{\phi,k} = \int_{-\infty}^{\infty} f(x)\phi^*(x-k)dx$$
 (3.29)

$$\widetilde{a}_{\phi,k} = \int_{-\infty}^{\infty} f(x)\phi(x-k)dx$$
(3.30)

In terms of these coefficients, the wavelet expansion of a general function f(x) for which

$$\int_{-\infty}^{\infty} |f(x)|^2 dx < \infty \qquad (3.31) \qquad \text{is given by:}$$

$$f(x) = \sum_{k=-\infty}^{\infty} (a_{\phi,k} \phi(x) + \widetilde{a}_{\phi,k} \phi^*(x-k)) + \sum_{j=0}^{\infty} \sum_{k=\infty}^{\infty} (a_{j,k} w(2^j x - k) + \widetilde{a}_{j,k} w^*(2^j x - k)) (3.32)$$

Now we try to find out an algorithm to compute the coefficients $a_{\phi,k}$, $\tilde{a}_{\phi,k}$, $a_{j,k}$, $\tilde{a}_{j,k}$ in this expansion. As an example, consider $a_{j,k}$ in (3.27). First substitute for $w^*(2^j x - k)$ in terms of its Fourier transform where from (3.15)

$$w^{*}(2^{j}x-k) = \int_{2\pi 2^{j}}^{4\pi 2^{j}} (\frac{1}{2\pi}) 2^{-j} e^{\frac{i\omega k}{2^{j}}} e^{-i\omega x} d\omega$$
(3.33)

So that $a_{j,k}$ in (2.27) becomes

$$a_{j,k} = \frac{1}{2\pi} \int_{2\pi 2^{j}}^{4\pi 2^{j}} d\omega e^{i\omega k/2^{j}} \int_{-\infty}^{\infty} dx f(x) e^{-i\omega x} dx$$
(3.34)

Since the Fourier transform of $w^*(2^j x - k)$ is zero outside $2\pi 2^j \le w < 4\pi 2^j$ and second

term integral over x is just the Fourier transform of f(x) multiplied by 2π so that

$$a_{j,k} = \int_{2\pi 2^{j}}^{4\pi 2^{j}} F(\omega) e^{\frac{i\omega k}{2^{j}}} d\omega$$
(3.35)

Now we replace integral by summation to get the discrete form of the wavelet transform. In this case, we step samples in increment of 2π in frequency band so that there 2^{j} steps in level *j*. Then $F(\omega)$ is replaced by the discrete coefficient $F_{2^{j}+s}$ when $F_{2^{j}+s} = 2\pi F(\omega = 2\pi (2^{j} + s)).$

And the integral in (3.35) becomes the summation

$$a_{2^{j}+k} = \sum_{s=0}^{2^{j}-1} F_{2^{j}+s} e^{i2\pi \left(2^{j}+s\right)k} / 2^{j}$$
(3.36)

Because $\nabla \omega = 2\pi$ cancels with $\frac{1}{2\pi}$ from (3.27) and $e^{i2\pi k} = 1$ for all integer k we obtain the formula:

$$a_{2^{j}+k} = \sum_{s=0}^{2^{j}-1} F_{2^{j}+s} e^{i2\pi s k/2^{j}} \quad k=0 \text{ to } 2^{j}-1$$
(3.37)

This is the inverse discrete Fourier transform for the sequence of frequency coefficients $F_{2^{j}+s}$, s=0 to $2^{j}-1$.

To compute the wavelet amplitude coefficients by a discrete algorithm, we begin by representing f(x) by the discrete sequence f(r), r=0 to N-1(where N is power of 2). Next FFT is used to compute the set of complex frequency coefficients F(p), p=0 to N-1, then octave blocks of F(p) are processed by IFFT to generate the amplitudes of the harmonic wavelet expansion of f(r) with the coefficients:

$$\widetilde{a}_{2^{j}+k} = \sum_{s=0}^{2^{j}-1} F_{-(2^{j}+s)} e^{-i2\pi s k/2^{j}} \quad k=0 \text{ to } 2^{j} -1$$
(3.38)

Because the discrete Fourier transform does not have negative indices but instead of F_{-s} by F_{N-S} that becomes:

$$\widetilde{a}_{2^{j}+k} = \sum_{s=0}^{2^{j-1}} F_{N-(2^{j}+s)} e^{-i2\pi s k/2^{j}}$$
(3.39)

As an example, for N=16, Figure 3.16 shows how the algorithm works in diagram. For (a) f(r) is a real function and $\tilde{a}_{2^{j}+k}$ are complex conjugates of $a_{2^{j}+k}$. It is not necessary to calculate these coefficients. And in Figure 3.17 for (b) f(r) is complex as shown. And the computation of a_0 and $a_{N/2}$ in algorithm involves special cases.

First, for a_0 consider

$$\phi(x-k) = \int_{-\infty}^{\infty} \phi(\omega) e^{-i\omega k} e^{i\omega x} d\omega$$
(3.40)

$$a_{\phi,k} = \int_{-\infty}^{\infty} dx \int_{-\infty}^{\infty} d\omega \phi^*(\omega) e^{i\omega k} f(x) e^{-i\omega x} = \int_{0}^{2\pi} d\omega F(\omega) e^{i\omega k}$$
(3.41)

Because $\phi(\omega)$ is zero except for $0 \le \omega < 2\pi$ when it is $\frac{1}{2\pi}$, the integral is covered by a single of 2π . If the value of $F(\omega=0)$ is replaced by F_0 , then

$$F_0 = 4\pi F(\omega = 0), a_{\phi,0} = \frac{F_0}{2}, \widetilde{a}_{\phi,0} = \frac{F_0}{2}$$
. Hence, $a_0 = a_{\phi,0} + \widetilde{a}_{\phi,0} = F_0$.

Second, for a_{N_2} , the initial FFT operation which transform f(r) to F(p) preserves the mean square so that $\sum_{p=0}^{N-1} |F(p)|^2 = \frac{1}{N} \sum_{r=0}^{N-1} |f(r)|^2$ holds. Each subsequent IFFT operation on octave bands of F(p) for p<N/2 or its complex conjugate on octave band for p>N/2 similarly preserves the mean square in octave bands so that $\frac{1}{2^j} = \sum_{k=0}^{2^j-1} |a_{2^{j+k}}|^2 = \sum_{k=0}^{2^j-1} |F_{2^{j+k}}|^2$ holds. Hence one has $a_{N_2} = F_{N_2}$.

The Parseval's theory in form is:

$$\left|a_{0}\right|^{2} + \sum_{j=0}^{N-2} \frac{1}{2^{j}} \sum_{k=0}^{2^{j}-1} \left(\left|a_{2^{j}+k}\right|^{2} + \left|a_{N-2^{j}-k}\right|^{2}\right) + \left|a_{N/2}\right|^{2} = \frac{1}{N} \sum_{r=0}^{N-1} \left|f(r)\right|^{2}$$
(3.42)

The summation is made over all levels j=0 to n-2, the highest level n-2 being the maximum that can reach with initial sequence of length 2^n . For example an original sequence length of $2^4 = 16$ can provide harmonic wavelet amplitudes at level -1, 0, 1, and n-2=2, comparing with the levels that obtained from the dilation wavelet, the levels from -1, 0, 1, 2 and n-1=3. Since for each pair of j, k, the dilation method has only one real dilation wavelet whereas harmonic wavelet has two real ones that are even and odd which are given by the real and imaginary parts of the complex harmonic wavelet [49].

The algorithm illustrated in Figure 3.16 is a fast way to compute the discrete harmonic wavelet transform and for the inverse transform, the same algorithm works in reverse

starting at the bottom of diagrams in Figure 3.17.

In summary, the wavelet determined from the dilation equation is a sequential algorithm in the sense that successive wavelets levels are calculated in sequence. In contrast, the FFT algorithm for harmonic wavelets are parallel method that FFT is only used once to get full sequence and then in the second time in octave block all of which are computed simultaneously. Therefore, for application which is involved very long sequence or complicate signal processing, harmonic wavelet transform algorithm is a very significant time-saving method.



Figure 3. 16 FFT algorithm to compute the harmonic wavelet transform: (a) for a

sequence of 16 real elements;[46]


Figure 3. 17 FFT algorithm to compute the harmonic wavelet transform: (b) for a

sequence of 16 complex elements [46]

3.5 Conclusion

In this chapter, the mechanism of a signal decomposition and reconstruction is presented based on the wavelet filter processing. The properties and characteristics of some typical harmonic wavelet transform are discussed with emphasis on two types of wavelet basis functions i.e. dilation wavelet function and harmonic wavelet function. Comparing these two wavelet functions, the conclusion is drawn that the harmonic wavelet function is more suitable to be the basis function in the wavelet-based linearization method which will be introduced in Chapters 4 & 5.

4 WAVELET-BASED LINEARIZATION FOR DUFFING OSCILLATOR SYSTEM

In this chapter, we will first introduce one kind of nonlinear system-Duffing oscillator system with non-linear stiffness. A multi-frequency linearization and wavelet-based linearization methods are implemented and compared for Duffing oscillator system when the system is subject to multiple harmonic inputs.

4.1 Duffing Oscillator System Model

To illustrate the procedure of equivalent linearization theory, first we consider the following oscillator with non-linear stiffness [54], as shown in Figure 4.1:



Non-linear stiffness

Figure 4. 1 SDOF mass-spring-damper system [54]

The ordinary differential equation of motion can be written as:

$$m\ddot{x}(t) + c\dot{x}(t) + g(x) = F(t) \tag{4.1}$$

where m is the mass, c is the viscous damping coefficient, F(t) is the external excitation,

and x(t) is the displacement response of system.

Dividing the equation by *m*, the equation becomes:

$$\ddot{x}(t) + \beta \dot{x}(t) + h(x) = f(t) \tag{4.2}$$

where β is the damping parameter, h(x) is the non-linear restoring force that could depend on displacement, and f(t) is a zero mean stationary random excitation.

We can always find a way to decompose the non-linear restoring force to one linear component plus a non-linear component, that is:

$$h(x) = \omega_n^2 \left[x + \lambda H(x) \right] \tag{4.3}$$

where λ is the non-linear factor that presents the type and degree of non-linearity in the system, and ω_n is the un-damped natural frequency for linear system.

The idea of linearization is replacing the equation (4.2) by the following linear system:

$$\ddot{x} + \beta_{eq}\dot{x} + \omega_{eq}^2 x = f(t) \tag{4.4}$$

where β_{eq} is the equivalent linear damping coefficient per unit mass, ω_{eq}^2 is the equivalent linear stiffness coefficient per unit mass, and $\beta_{eq} = \frac{\omega_n}{\omega_{eq}}\beta$, to find an

expression for ω_{eq}^2 is to minimize the expected value of the mean square error which is:

$$\varepsilon = h(x) - \omega_{eq}^2 x \tag{4.5}$$

The way that minimizes the expected value of mean square error, is shown:

$$\frac{\partial}{\partial \omega_{eq}^2} E[\varepsilon^2] = 0 \tag{4.6}$$

By substituting (4.5) into equation (4.6), and carrying out the differentiation, the following equations for ω_{eq}^2 is shown as:

$$\omega_{eq}^{2} = \frac{E\{xh(x)\}}{E\{x^{2}\}} = \frac{E\{xh(x)\}}{\sigma_{x}^{2}}$$
(4.7)

where σ_x is the standard deviation of x(t). By substituting the expression for h(x), given by (4.3), into equation(4.7), the result is:

$$\omega_{eq}^{2} = \omega_{n}^{2} \left(1 + \lambda \frac{E\{xH(x)\}}{\sigma_{x}^{2}} \right)$$
(4.8)

This expression shows, very clearly, how the non-linear component of the stiffness element influences the value of ω_{eq}^2 . In equation (4.8), the exact evaluation of

 $E\left\{\frac{xH(x)}{\sigma_x^2}\right\}$ requires a knowledge of the first order density function of the response

process x(t) which is unknown [54]. However the displacement x(t) is assumed to be Gaussian and Equation (4.7) becomes:

$$\omega_{eq}^{2} = E\left\{\frac{\partial h}{\partial x}\right\} = \omega_{n}^{2}\left(1 + \lambda E\left\{\frac{dH}{dx}\right\}\right)$$
(4.9)

For example, the Duffing oscillator is used to illustrate this procedure by which the non-linear restoring force is written as:

$$h(x) = \omega_n^2 \left(x + \lambda x^3 \right) \tag{4.10}$$

In this case, ω_{eq}^2 can be expressed:

$$\omega_{eq}^2 = \omega_n^2 \left(1 + 3\lambda E \left\{ x^2 \right\} \right) \tag{4.11}$$

And by defining:

$$A = E\left\{x^2\right\} \tag{4.12}$$

Then assume the density function of x(t) to be the Gaussian form:

$$f_x(x) = \frac{1}{(2\pi)^{\frac{1}{2}}\sigma_x} \exp\left(-\frac{x^2}{2\sigma_x^2}\right)$$
(4.13)

To combine (4.12) and (4.13) together, one has

$$A = E(x^{2}) = \int_{-\infty}^{\infty} x^{2} f_{x}(x) dx$$
(4.14)

And by performing integration the equation (4.14) with the Gamma function, the result is obtained:

$$A = \frac{2}{\sqrt{\pi}} \sigma_x^2 \Gamma\left(\frac{3}{2}\right) = \sigma_x^2 \tag{4.15}$$

The next step is to evaluate σ_x and the procedure is to use the frequency domain. The input-output formula $S_f(\omega)$ is the spectral density for f(t) and $S_x(\omega)$ is the spectral density for output x(t). Use the well-known relationship in the linear vibration :

$$S_{x}(\omega) = |\alpha(\omega)|^{2} S_{f}(\omega)$$
(4.16)

where $\alpha(\omega)$ is the frequency response function given as:

$$\alpha(\omega) = \frac{1}{\left(\omega_{eq}^2 - \omega^2 + 2i\xi_{eq}\omega\omega_{eq}\right)} \tag{4.17}$$

Once $S_x(\omega)$ is determined, σ_x is found from the equation:

$$\sigma_x^2 = \int_{-\infty}^{\infty} S_x(\omega) d\omega \tag{4.18}$$

The equations (4.16), (4.17), and (4.18) yield a relationship between ω_{eq} and σ_x . Another independent relationship between two quantities can be found from Equations (4.11) (4.14), (4.13) and (4.15). Hence, two algebraic relationship are obtained for two unknowns, ω_{eq} and σ_x , and the formulas for ω_{eq} and σ_x expression are formed. By iterative process the equivalent coefficient finally is computed. The procedure can be described as follows:

- 1. Assign an initial value of equivalent coefficient ω_{eq} .
- 2. Use equation (4.16), (4.17), (4.18) to get σ_x .
- 3. Solve equation (4.11) and (4.14) for the new ω_{eq} .

4. Repeat step 2 and step 3 until the new ω_{eq} converge to the old ω_{eq} within a pre-defined threshold.

4.2 Multi-Frequency Linearization for Stationary Inputs

In this section, we will discuss the multi-frequency-linearization method based on the equivalent linearization procedure and use Duffing system as an example.

For this kind of system, the non-linear factor only depends on the displacement so that the nonlinearity restoring force can be decomposed by linear one term and another non-linear one that mentioned in previous part. That is:

$$h(x) = kx + h_{nl}(x)$$
 (4.19)

The equivalent method is to replace the original non-linear system given by:

$$\ddot{x} + \beta \dot{x} + h(x) = f(t) \tag{4.20}$$

with the linear system given by

$$\ddot{x} + \beta \dot{x} + kx + k_{nl}x = f(t) \tag{4.21}$$

where k_{nl} is the equivalent coefficient which needs to be found out. Following the method proposed by Caughey [13][14], it is assumed that the damping β is very small and f(t) is a wide-band process. Under these circumstances the solution of equation (4.21) is the narrow-band process and one can write

$$x(t) = A(t)\cos(\omega_{eq}t + \theta(t))$$
(4.22)

$$x(t) = -\omega_{eq} A(t) \sin(\omega_{eq} t + \theta(t))$$
(4.23)

where
$$\omega_{eq}^2 = k + k_{nl}$$
 (4.24)

When β is small, A(t) and $\theta(t)$ will be slowly varying function of time over one cycle of response in period of $T = \frac{2\pi}{\omega_{eq}}$. Thus the expected value in the equation (4.7)

can be treated as average value over one cycle. On minimizing the mean square of error one cycle of response, k_{nl} is obtained as:

$$\omega_{eq}^{2} = \frac{\oint h(x)xdx}{\oint x^{2}dx}$$
(4.25)

For Duffing oscillator, the non-linear term is given by

$$h_{nl}(x) = k_{nl}x = \lambda kx^3 \tag{4.26}$$

And equation (4.20) becomes:

$$\ddot{x} + \beta \dot{x} + kx + \lambda kx^3 = f(t) \tag{4.27}$$

And the equivalent stiffness is obtained according to equation (4.25)

$$k_{nl} = \frac{\lambda k \int_0^T x^4 dx}{\int_0^T x^2 dx}$$
(4.28)

From the linear vibration theory, the input signal is a sum of sinusoids:

$$f(t) = F_1 \sin(\omega_1 t) + F_2 \sin(\omega_2 t) + \dots + F_N \sin(\omega_N t)$$
(4.29)

An then the output will be:

. . .

$$x(t) = X_1 \sin(\omega_1 t + \phi_1) + X_2 \sin(\omega_2 t + \phi_2) + \dots + X_N \sin(\omega_N t + \phi_N)$$
(4.30)

where
$$X_i^2 = \frac{F_i^2}{(k - \omega_i^2)^2 + (\beta \omega_i)^2}$$
 and $\phi_i = \tan^{-1} \frac{\beta \omega_i}{k - \omega_i^2}$ (4.31)

Substituting x(t) from (4.30) into (4.28), the additional stiffness due to the nonlinearity found for each frequency of excitation is calculated as

$$k_{nl,1} = \frac{3}{4}\lambda k X_1^2 \left[1 + 2\left(\frac{X_2}{X_1}\right)^2 + 2\left(\frac{X_3}{X_1}\right)^2 + \dots + 2\left(\frac{X_N}{X_1}\right)^2 \right]$$
$$k_{nl,2} = \frac{3}{4}\lambda k X_2^2 \left[2\left(\frac{X_1}{X_2}\right)^2 + 1 + 2\left(\frac{X_3}{X_2}\right)^2 + \dots + 2\left(\frac{X_N}{X_2}\right)^2 \right]$$

70

$$k_{nl,N} = \frac{3}{4} \lambda k X_N^2 \left[2 \left(\frac{X_1}{X_N} \right)^2 + 2 \left(\frac{X_2}{X_N} \right)^2 + 2 \left(\frac{X_3}{X_N} \right)^2 + \dots + 1 \right]$$
(4.32)

To obtain the solution for this equation, we need the iterative scheme. Starting with

$$k_{nl,i} = 0$$
 and $X_i = \frac{F_i}{\sqrt{(-\omega_i^2 + k + k_{nl,i})^2 + (\beta\omega_i)^2}}$ for i=1:N (4.33)

The coefficient k_{nl} is computed by equation (4.32) until it is converged and the result for above system only produces the same frequency as the excitation one. It has been shown that (Stoker 1965) [63] when the harmonic solution is known, the response of the Duffing oscillator to a single frequency can be expressed by

$$x_i(t) = X_i \sin(\omega_i t) - \frac{k\lambda X_i^3}{36\omega_i^2} \sin(3\omega_i t) + \frac{k\lambda^2 X_i^5}{1296\omega_i^4} \sin(5\omega_i t)$$
(4.34)

The whole response can be calculated by summing up all the response components as

$$x(t) = \sum_{i=1}^{N} x_i(t)$$
(4.35)

The example is given to display this procedure. The system is:

$$\ddot{x} + \beta \dot{x} + kx + \lambda kx^3 = f(t)$$
, $x(0) = \dot{x}(0) = 0$, $\beta = 0.5, k = 16\pi^2, \lambda = 1$ and
 $f(t) = 15\sin(4t) + 10\sin(8t) + 5\sin(12t) + 5\sin(16t)$ (4.36)

•

The input and output process are shown as in Figure 4.2



Figure 4. 2 Equivalent linearization for stationary input

The Figure 4.2 shows that the linearization method yields a nice approximation of the original system response because of the stationary excitation. In the next section, a wavelet-based linearization method is proposed for Duffing oscillator.

4.3 Wavelet-Based Linearization for Duffing Oscillator

Although the method of equivalent linearization to analyze the non-linear system response is quite successful in engineering applications, it is still limited at some aspects that the response must be harmonic in nature. Thus if the response energy fluctuates temporally, a time –frequency representation may approximate the response of the system better.

Since the wavelet analysis provide a powerful way for response of system subjected to the non-stationary signal, especially, wavelet transformation can be used to obtain the time-frequency local information of functions very well, to develop a wavelet-based method might be a good alternative to traditional equivalent linearization method.

In this section we introduce the harmonic wavelet function as the basis function to develop the wavelet-based linearization method.

The harmonic wavelet has some typical features different with others. First, it has explicit form function and the definition of a harmonic wavelet at a particular scale is quite easy, unlike the dilation wavelets that must be calculated by iterative process. Second, due to its compact supported form in frequency domain that builds up the non-overlapping frequency bands to form an orthogonal basis, the level of a wavelet is interchangeable with its frequency band. And also because of its frequency characteristic, the fast algorithm can be implemented to reduce the burden of computation.

As any function can be expressed by a series of linear combination components with decomposed scales and positions of harmonic wavelet functions, one has:

$$x(t) = \sum_{j=0}^{n-2} x_j(t)$$
(4.37)

with

$$x_{j}(t) = \sum_{k=0}^{2^{2}-1} a_{j,k} w(2^{j}t - k) + a_{j,k}^{*} w^{*}(2^{j}t - k)$$
(4.38)

where the symbol * denotes complex conjugate. For zero mean and real value x(t)

$$x_{j}(t) = 2 \operatorname{Re}\left[\sum_{k=0}^{2^{j}-1} a_{j,k} w(2^{j}t - k)\right]$$
(4.39)

where Re[] represents the real part of a complex number. The harmonic wavelet coefficients are defined as [46],

$$a_{j,k} = \int_{-\infty}^{\infty} x(t) w^* (2^j t - k) dt$$
(4.40)

In the frequency domain, equation(4.40) becomes

$$A_{j,k}(\omega) = W_{j,k}(\omega)X(\omega) \text{ for } 2^{j}2\pi \le \omega < 2^{j}4\pi$$
$$A_{j,k}(\omega) = 0, \text{ else.}$$
(4.41)

For the level *j* and translated by *k* steps of size $\frac{1}{2^j}$, the wavelet transform is:

$$W_{j,k} = \left(\frac{1}{2\pi}\right) 2^{-j} e^{-i\omega k/2^{j}} \text{ for } 2^{j} 2\pi \le \omega < 2^{j} 4\pi$$

$$W_{j,k} = 0 \text{, elsewhere}$$
(4.42)

And for a discrete sequence x_n , n=1 to N-1, discrete Fourier coefficients of harmonic wavelet transform for level *j* can be expressed as

$$A_{k} = X_{k} W_{k}^{*} \text{ for } k = 0 to 2^{j} - 1$$
(4.43)

where X_k for $k = 0to2^j - 1$ are the Fourier coefficients of x_n for n=1 to N-1, and the square of the absolute values are:

$$A_k^2 = X_k^2 W_k^2 \quad \text{for} \quad k = 0 \text{ to } 2^j - 1 \tag{4.44}$$

From equation (4.42), the absolute value of harmonic wavelet for scale j is:

$$|W_{j,k}(\omega)| = \frac{1}{2\pi} 2^{-j} \qquad 2^{j} 2\pi \le \omega < 2^{j} 4\pi$$

$$|W_{j,k}(\omega)| = 0, \text{ else.} \qquad (4.45)$$

Substituting equation (4.45) into (4.44), a relationship between the absolute values of Fourier coefficients of response and its wavelet coefficients is clearly displayed.

$$A_{k}^{2} = \frac{1}{4\pi^{2}(2^{j})^{2}} X_{k}^{2} \qquad \text{for } k = 0to2^{j} - 1.$$

$$A_{k}^{2} = 0, \text{ else.}$$
(4.46)

Or changed to:

$$X_{k}^{2} = 4\pi^{2} (2^{j})^{2} A_{k}^{2} \qquad \text{for } k = 0 to 2^{j} - 1.$$

$$X_{k}^{2} = 0, \text{ else.} \qquad (4.47)$$

Using equation (4.47) in equation (4.32), the scale dependent equivalent stiffness values are calculated as

$$k_{nl,0} = 3\pi^{2} \left(2^{j}\right)^{2} \lambda k A_{2^{j}}^{2} \left[1 + 2 \left(\frac{A_{2^{j}+1}}{A_{2^{j}}}\right)^{2} + 2 \left(\frac{A_{2^{j}+2}}{A_{2^{j}}}\right)^{2} + \dots + 2 \left(\frac{A_{2^{j}+k}}{A_{2^{j}}}\right)^{2} \right]$$
$$k_{nl,1} = 3\pi^{2} \left(2^{j}\right)^{2} \lambda k A_{2^{j}+1}^{2} \left[2 \left(\frac{A_{2^{j}}}{A_{2^{j}+1}}\right)^{2} + 1 + 2 \left(\frac{A_{2^{j}+2}}{A_{2^{j}+1}}\right)^{2} + \dots + 2 \left(\frac{A_{2^{j}+k}}{A_{2^{j}+1}}\right)^{2} \right]$$

• • • •

$$k_{nl,k} = 3\pi^2 \left(2^j\right)^2 \lambda k A_{2^j+k}^2 \left[2\left(\frac{A_{2^j}}{A_{2^j+k}}\right)^2 + 2\left(\frac{A_{2^{j+1}}}{A_{2^{j+k}}}\right)^2 + 2\left(\frac{A_{2^{j+2}}}{A_{2^{j+k}}}\right)^2 + \dots + 1 \right]$$
(4.48)

and the transfer function is updated. The equation (4.48) expresses that for the level *j*, there are 2^{j} equivalent stiffness coefficients, each $\frac{1}{2^{j}}$ units apart on the time axis, the process is iterated until the solution converges.

The wavelet-based linearization results are displayed and compared with numerical method of the 4th Runge-Kutta as shown in Figure 4.3.



Figure 4. 3 Wavelet-based linearization solution

For a given system and the input signal with the length N, the response is obtained by using the Fourier transform of the excitation multiplying the frequency response function but ignoring the non-linearity. Next, every level (scale) of the wavelet coefficient of the response in the frequency domain is calculated with the Newland dyadic harmonic [46] wavelet as the basis function. At last, the equivalent parameters are obtained when the iterative process converges.

In order to illustrate the procedure validity, the wavelet-based linearization method is applied to the system defined in equation (4.36). Figure 4.2 shows the excitation and system response by using equivalent linearization method. Figure 4.3 shows the response of non-linear system using the wavelet-based method comparing with the numerical solution of ordinary differential equation using 4th order Runge-Kutta algorithm. From figures, the wavelet-based algorithm developed above yields a good approximation of the exact response.

In summary, the statistical linearization method provides the way that obtains the parameters in the equivalent linear model by minimizing the mean square of the equation deficiency, but this procedure needs the knowledge of the probability distribution function. In order to avoid the complication of computing, an alternative approach is required called multi-frequency linearization method with some assumptions, the parameters are obtained by iterative algorithm. Since the harmonic wavelets function is compact in the frequency domain, its Fourier transform offers explicit form to the wavelet coefficients with the algebraic transformation of the response in frequency domain. The equivalent linearization parameters are obtained in both time and

frequency domain by using wavelet-based linearization technique.

4.4 Conclusion

This chapter presents the procedure of linearzing the Duffing oscillator system by using traditional and wavelet-based equivalent linearization methods. First of all, it introduces the statistical linearization method and then extends to the multi-frequency approach which is based on this statistical method to derive the equivalent linearization parameters in this Duffing system. The parameters can be determined by multi-frequency Fourier transform of responses according to the multi-frequency system excitations respectively. By using the harmonic wavelet basis function which provides the non-over lapping frequency bands and scale, a wavelet-based linearization method for the Duffing system is proposed and the simulation results are compared with the numerical method Runge-Kutta to show that the proposed linearization method is an appealing approach to linearize the nonlinear Duffing systems.

5 WAVELET-BASED LINEARIZATION FOR BILINEAR HYSTERETIC SYSTEMS

In this chapter, the wavelet-based linearization method is developed for another type of nonlinear system- bilinear hysteresis system. The organization of this chapter is: Section 5.1 introduces bilinear hysteresis system model. Section 5.2 describes the statistic linearization equivalent method for the bilinear hysteresis system. The basic concept of wavelet transform for the relationship of input and output in the linear system is introduced in Section 5.3. In the Section 5.4, a wavelet-based linearization method is proposed to obtain equivalent linear parameters based on both frequency and time information of the nonlinear system. The numerical results are provided in Section 5.5.

5.1 Bilinear Hysteretic System Model

For the system of bilinear hysteresis, the system model is

$$\ddot{x} + \beta \dot{x} + h(\dot{x}, x, t) = f(t) \quad x(0) = \dot{x}(0) = 0$$
(5.1)

where the hysteretic force is

$$h(x, \dot{x}, t) = \alpha x + (1 - \alpha) h_0(\dot{x}, y; x_y)$$
(5.2)

$$\dot{y} = h_1(\dot{x}, y; x_y) \tag{5.3}$$

where $h_0 = y \{ u(y + x_y) - u(y - x_y) \} + x_y \{ u(y - x_y) u(\dot{x}) - u(-y - x_y) u(-\dot{x}) \}$

$$h_{1} = \dot{x} \{ u(y + x_{y}) - u(y - x_{y}) + u(y - x_{y})u(-\dot{x}) + u(-y - x_{y})u(\dot{x}) \}$$
(5.4)

where y is the relative displacement of the purely elasto-plastic component h_0 , the hysteresis force of $h(x, \dot{x}, t)$ is assumed to have the characteristic as shown in Figures 5.1-5.3, $\alpha = 0.5$ and $x_y = 1$ are respectively the second slope ratio and the yield displacement, u() is a unit step function.

The bilinear hysteresis and elasto-plastic hysteresis loops are shown in Figures 5.4 and 5.5.



Figure 5. 1 The bilinear system model



Figure 5. 2 The bilinear hysteresis force h characteristic $\alpha = \tan \varphi$



Figure 5. 3 Elasto-plastic hysteresis h_0



Figure 5. 4 The bilinear hysteresis loop



Figure 5. 5 The elaso-plastic hysteresis loop

The equivalent linearization model to the system is given by the formula:

$$\ddot{x} + \beta \dot{x} + \alpha x + (1 - \alpha)(c_1 \dot{x} + c_2 x) = f(t)$$
(5.5)

$$\dot{y} = c_3 \dot{x} + c_4 y \tag{5.6}$$

Equation (5.5) can be written as:

$$\ddot{x} + (\beta + c_1(1 - \alpha))\dot{x} + (\alpha + (1 - \alpha)c_2)x = f(t)$$
(5.7)

$$\ddot{x} + \beta_{eq} \dot{x} + \omega_{eq}^2 x = f(t) , (5.7a) \quad \beta_{eq} = (\beta + c_1(1 - \alpha)), (5.7b) \quad \omega_{eq}^2 = (\alpha + (1 - \alpha)c_2)(5.7c)$$

where β_{eq} and ω_{eq}^2 are the equivalent linearized parameters.

5.2 Statistical Linearization for Bilinear Hysteretic System

The statistical linearization technique is used to investigate this proposal method [58], [59]. Consider the same bilinear hysteresis system given by

$$\ddot{x} + \beta \dot{x} + h(x, \dot{x}, t) = f(t) \quad x(0) = \dot{x}(0) = 0$$
(5.8)

where
$$\beta = 0.02$$
 and $f(t) = 15\sin(4t) + 10\sin(8t) + 5\sin(12t) + 5\sin(16t)$ (5.9)

The hysteretic force $h(x, \dot{x}, t) = \alpha x + (1 - \alpha) h_0(\dot{x}, y; x_y)$ (5.10)

$$\dot{y} = h_1(\dot{x}, y; x_y)$$
 (5.11)

where $h_0 = y \{ u(y + x_y) - u(y - x_y) \} + x_y \{ u(y - x_y) u(\dot{x}) - u(-y - x_y) u(-\dot{x}) \}$

$$h_{1} = \dot{x} \{ u(y + x_{y}) - u(y - x_{y}) + u(y - x_{y})u(-\dot{x}) + u(-y - x_{y})u(\dot{x}) \}$$
(5.12)

where $\alpha = 0.5$ and $x_y = 1$ are respectively the second slope ratio and the yield displacement, u() is a unit step function. The numerical method is used to obtain solution of h_0 and displacement of x.

Meanwhile, we use the statistical linearization method [7] that replaces the nonlinear system in equation (5.8) with the linear one as:

$$\ddot{x} + \beta \dot{x} + \alpha x + (1 - \alpha)(C_1 \dot{x} + C_2 x) = f(t)$$

$$\dot{y} = C_3 \dot{x} + C_4 y$$
(5.13)

where the equivalent coefficients C_1, C_2, C_3, C_4 can be calculated by assuming a Gaussian probability distribution of the multi-dimensional response process which are

$$C_{1} = E\begin{bmatrix}\frac{\partial h_{0}}{\dot{x}}\end{bmatrix} = \frac{x_{y}}{\sqrt{2\pi\sigma_{\dot{x}}}} \operatorname{erfc}\left(\frac{x_{y}}{\alpha_{y}\sqrt{2(1-\rho^{2})}}\right)$$

$$C_{2} = E\begin{bmatrix}\frac{\partial h_{0}}{y}\end{bmatrix} = \operatorname{erf}\left(\frac{x_{y}}{\sqrt{2\sigma_{y}}}\right) - \frac{x_{y}}{\sqrt{2\pi\sigma_{y}}} \exp\left(\frac{-x_{y}^{2}}{2\sigma_{y}^{2}}\right) \operatorname{erfc}\left(\frac{\rho x_{y}}{\sigma_{y}\sqrt{2(1-\rho^{2})}}\right)$$

$$C_{3} = E\begin{bmatrix}\frac{\partial h_{1}}{\dot{x}}\end{bmatrix} = \frac{1}{2}\left\{1 + \operatorname{erf}\left(\frac{x_{y}}{\sqrt{2\sigma_{y}}}\right)\right\} - \frac{1}{2\pi}\exp\left(\frac{-\rho x_{y}^{2}}{2\sigma_{y}^{2}\sqrt{(1-\rho^{2})}}\right)$$

$$C_{4} = E\begin{bmatrix}\frac{\partial h_{1}}{\partial y}\end{bmatrix} = -\frac{\rho x_{y}\sigma_{\dot{x}}}{\sqrt{2\pi\sigma_{y}^{2}}} \exp\left(\frac{-x_{y}^{2}}{2\sigma_{y}^{2}}\right)\left[1 + \operatorname{erf}\left(\frac{\rho x_{y}}{\sigma_{y}\sqrt{2(1-\rho^{2})}}\right)\right] - \frac{\sigma_{\dot{x}}\sqrt{(1-\rho^{2})}}{\pi\sigma_{y}}\exp\left(-\frac{x_{y}^{2}}{2(1-\rho^{2})\sigma_{y}^{2}}\right)\right]$$
(5.14)

In which $\sigma_{\dot{x}}$ and σ_{y} are respectively the root mean square value of \dot{x} , the parameters y, ρ are the coefficient of correlation of \dot{x} and y, and erf() and erf() are respectively the error function and complementary error function.

5.3 Input - Output Relationship for a Linear System

Since the wavelet transform is the convolution operation in time, where the signal to the wavelet transform is decomposed in the distinct scales and translation values, the wavelet coefficients calculation is quite similar to the calculation of the response of linear system that the convolution operation is implemented using impulse response function in the LTI system. In the LTI system, the response calculation in the wavelet basis is obtained to replace the impulse function with the wavelet function.

Consider a linear SDOF system with the equation motion:

$$\ddot{x}(t) + \beta \dot{x}(t) + kx(t) = f(t)$$
(5.15)

where β is damping parameter, k is stiffness parameter. By performing the wavelet transform both sides of (5.15), one has [9], [10]:

$$W_{\varphi}\ddot{x}(a,b) + \beta W_{\varphi}\dot{x}(a,b) + kW_{\varphi}x(a,b) = W_{\varphi}f(a,b)$$
(5.16)

where $W_{\phi}\ddot{x}, W_{\phi}\dot{x}, W_{\phi}x, W_{\phi}f$ denote the wavelet transform of \ddot{x}, \dot{x}, x, f respectively, performing the integration by part on $W_{\phi}\ddot{x}(a,b)$ according to the characteristic of fast decaying for wavelet basis, the following expression can be obtained:

$$W_{\varphi}\ddot{x}(a,b) = \frac{1}{a^2} W_{\varphi}x(a,b)$$
(5.17)

where the term on the right hand side is the wavelet transform with the wavelet basis $\ddot{\varphi}$, the second order partial differential equation of $W_{\varphi}x(a,b)$ with respect b can be expressed as:

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) = \frac{\partial^2}{\partial b^2} \int_{-\infty}^{\infty} x(t) \varphi\left(\frac{t-b}{a}\right) dt$$
(5.18)

Exchanging the differential operator with the integral operator on the right hand side the equation (5.18) can be written as

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) = \frac{1}{a^2} W_{\varphi} x(a,b)$$
(5.19)

Thus, one has

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) = W_{\varphi} \ddot{x}(a,b)$$
(5.20)

Similarly, the following is obtained

$$\frac{\partial}{\partial b}W_{\varphi}x(a,b) = W_{\varphi}\dot{x}(a,b)$$
(5.21)

The equation (5.16) can be expressed by: [8]

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) + \beta \frac{\partial}{\partial b} W_{\varphi} x(a,b) + k W_{\varphi} x(a,b) = W_{\varphi} f(a,b)$$
(5.22)

From this equation, the wavelet coefficient of the output can be obtained from the wavelet coefficient of the input. Because the parameter b contains time information, the information of response output in time domain is achieved as well.

It is noted that in the LTI linear system, the wavelet coefficients of response output calculation is obtained through the way that replaces impulse response function in time domain convolution operation in the response calculation by the wavelet basis function. Thus, we can have the response information about time-frequency combination.

5.4 Wavelet-Based Equivalent Linearization

Considering the bilinear system (5.1), by using the input-output relationship introduced previously and performing the wavelet transform on both sides of the equation (5.8) and

(5.7a), the following form can be obtained:[8][9]

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) + \beta \frac{\partial}{\partial b} W_{\varphi} x(a,b) + W_{\varphi} h(a,b) = W_{\varphi} f(a,b)$$
(5.23)

$$\frac{\partial^2}{\partial b^2} W_{\varphi} x(a,b) + \beta \frac{\partial}{\partial b} W_{\varphi} x(a,b) + \omega_{eq}^2 W_{\varphi} x(a,b) = W_{\varphi} f(a,b)$$
(5.24)

Following the basic procedure of the traditional equivalent linearization approach, one can first find the mean square least error between the equations (5.23) and (5.24)

$$\varepsilon^{2} = (\omega_{eq}^{2} W_{\varphi} x(a,b) - W_{\varphi} h(a,b))^{2}$$
(5.25)

In fact, the wavelet coefficients are calculated numerically. In this case, parameters a, b will be discretized as $a = 2^{j}$ and $b = k2^{j}$. By summing the discretized error over all j values and by multiplying by the factor $\frac{1}{2^{j}}$, which represents the error in the instantaneous energy of the response at a instant time interval k, then minimizing the square error with respect ω_{eq}^{2} , the instantaneous equivalent natural frequency will be obtained [62].

$$\frac{\partial}{\partial \omega_{eq,k}^2} \sum_{allj} \varepsilon_{j,k}^2 = 0$$
(5.26)

$$\omega_{eq,k}^{2} = \frac{\sum_{allj} W_{j,k} x W_{j,k} h}{\sum_{allj} W_{j,k}^{2} x}$$
(5.27)

Also by summing the discretized error for all the time positions over k which represents the frequency dependent equivalent parameter, one obtains

$$\frac{\partial}{\partial \omega_{eq,j}^2} \sum_{allk} \varepsilon_{j,k}^2 = 0$$
(5.28)

$$\omega_{eq,j}^{2} = \frac{\sum_{allk} W_{j,k} x W_{j,k} h}{\sum_{allk} W_{j,k}^{2} x}$$
(5.29)

Thus, the wavelet method can examine the error between the nonlinear and linearized system either at each time interval or each frequency band corresponding to wavelet scale. This flexible feature is very important to solve the nonlinear response of systems. One can compare with the same equivalent linearization model:

$$\ddot{x} + \beta_{eq}\dot{x} + \omega_{eq}^2 x = f(t)$$
(5.30)

with

$$\beta_{eq} = (\beta + c_1(1 - \alpha))(5.31), \text{ and } \omega_{eq}^2 = (\alpha + (1 - \alpha)c_2)$$
 (5.32)

The output x(t) and y(t) can be decomposed by using the harmonic wavelet transform to obtain:

$$x(t) = \sum_{j=0}^{n-2} x_j(t) \approx \sum_{j=0}^{n-2} \left(2 \operatorname{Re}\left(\sum_{k=0}^{2^j - 1} a_{j,k} w(2^j t - k) \right) \right)$$
(5.33)

where Re[] represents the real part of a complex number. The harmonic wavelet coefficients are defined as [46],

$$a_{j,k} = \int_{-\infty}^{\infty} x(t) w^* (2^j t - k) dt$$
(5.34)

Similarly,

$$y(t) = \sum_{j=0}^{n-2} y_j(t) \approx \sum_{j=0}^{n-2} \left(2 \operatorname{Re}\left(\sum_{k=0}^{2^j - 1} a'_{j,k} w(2^j t - k) \right) \right)$$
(5.35)

where

$$a'_{j,k} = \int_{-\infty}^{\infty} y(t) w^* (2^j t - k) dt$$
(5.36)

where $w(2^{j}t-k)$ is the harmonic wavelet transform function represented by the level (scale) j and time position of $k \cdot a_{j,k}$ and $a'_{j,k}$ are the wavelet coefficients of x(t) and y(t) respectively, also giving the assumption that x(t) and y(t) are both real functions.

In order to obtain the equivalent liner damping and stiffness parameters, also the equivalent linear parameters c_3 and c_4 , we use the method discussed previously that is called wavelet-based linearization method.

Performing the wavelet transform on both sides of the equations (5.9) and (5.30), the following equations can be obtained:

$$\frac{\partial^2}{\partial k^2} W_w x \left(2^j t - k \right) + \beta \frac{\partial}{\partial k} W_w x \left(2^j t - k \right) + W_w h \left(2^j t - k \right) = W_w f \left(2^j t - k \right)$$
(5.37)

$$\frac{\partial^2}{\partial k^2} W_w x \left(2^j t - k \right) + \beta_{eq} \frac{\partial}{\partial k} W_w x \left(2^j t - k \right) + \omega_{eq}^2 W_w x \left(2^j t - k \right) = W_w f \left(2^j t - k \right)$$
(5.38)

The square of error between the nonlinear and linearized formulation at each scale j and time position k of the wavelet transform is:

$$\mathcal{E}_{j,k}^{2} = \left[\beta_{eq} \frac{\partial}{\partial k} W_{w} x \left(2^{j} t - k\right) + \omega_{eq}^{2} W_{w} x \left(2^{j} t - k\right) - \beta \frac{\partial}{\partial k} W_{w} x \left(2^{j} t - k\right) - W_{w} h \left(2^{j} t - k\right)\right]^{2} (5.39)$$

The above error, when multiplying with reciprocal of the scale $\frac{1}{2^{j}}$ and summing over

all that scale values, represents the error in the instantaneous energy of the response at the time t=k. By minimizing this error of the instantaneous energy with respect to the equivalent damping parameter β_{eq} and stiffness parameter ω_{eq}^2 , the instantaneous equivalent damping and stiffness parameters of the linearization can be obtained.

Hence, a time variant linear system model is derived. The minimization conditions are:

$$\frac{\partial}{\partial \beta_{eq,k}} \sum_{allj} \frac{1}{2^j} \varepsilon_{j,k}^2 = 0$$
(5.40)

$$\frac{\partial}{\partial \omega_{eq,k}^2} \sum_{allj} \frac{1}{2^j} \varepsilon_{j,k}^2 = 0$$
(5.41)

Substituting equation (5.39) to (5.40) and (5.41) respectively, one has:

$$\beta_{eq,k} = \frac{\sum_{allj} \frac{\partial}{\partial k} W_w x(j,k) W_w h(j,k)}{\sum_{allj} \left(\frac{\partial}{\partial k} W_w x(j,k) \right)^2}$$
(5.42)
$$\omega_{eq,k}^2 = \frac{\sum_{allj} W_w x(j,k) W_w h(j,k)}{\sum_{allj} (W_w x(j,k))^2}$$
(5.43)

Similarly, when summing the error for all the time positions, the error represents the energy of the process at each frequency band corresponding to a wavelet scale. Thus, the equivalent damping and stiffness parameters of this linearized system are determined as shown:

$$\frac{\partial}{\partial \beta_{eq,j}} \sum_{allk} \varepsilon_{j,k}^2 = 0 \tag{5.44}$$

$$\frac{\partial}{\partial \omega_{eq,j}^2} \sum_{allk} \varepsilon_{eq}^2 = 0 \tag{5.45}$$

Thus, one has

$$\beta_{eq,j} = \frac{\sum_{allk} \frac{\partial}{\partial k} W_w x(j,k) W_w h(j,k)}{\sum_{allk} \left(\frac{\partial}{\partial k} W_w x(j,k) \right)^2}$$
(5.46)
$$\omega_{eq,j}^2 = \frac{\sum_{allk} W_w x(j,k) W_w h(j,k)}{\sum_{allk} (W_w x(j,k))^2}$$
(5.47)

For the equations (5.3) and (5.6), taking the wavelet transform on both sides of two equations, we obtain the formulations:

$$\frac{\partial}{\partial k}W_{w}y(2^{j}t-k) = W_{w}h_{1}(2^{j}t-k)$$
(5.48)

$$\frac{\partial}{\partial k}W_{w}y(2^{j}t-k) = c_{3}\frac{\partial}{\partial k}W_{w}x(2^{j}t-k) + c_{4}W_{w}y(2^{j}t-k)$$
(5.49)

Similarly, taking the square of error between the equation (5.48) and (5.49) at every scale j and time position k of their wavelet basis function, one obtains:

$$\varepsilon_{j,k,y}^{2} = \left[c_{3}\frac{\partial}{\partial k}W_{w}x\left(2^{j}t-k\right)+c_{4}W_{w}y\left(2^{j}t-k\right)-W_{w}h_{1}\left(2^{j}t-k\right)\right]^{2}$$
(5.50)

By using the same technique of finding the equivalent damping and stiffness parameters to obtain the parameters C_3 and C_4 under the minimization conditions, we obtain the following formulations:

$$\frac{\partial}{\partial c_{3,k}} \sum_{allj} \frac{1}{2^j} \varepsilon_{j,k,y}^2 = 0$$
(5.51)

$$\frac{\partial}{\partial c_{4,k}} \sum_{allj} \frac{1}{2^{j}} \varepsilon_{j,k,y}^{2} = 0$$
(5.52)

These two formulations are defined by summing the square of errors over all the scale values j to represent the energy of the process at the instantaneous k, so that the information about linear time variant systems are obtained with equivalent parameters C_3 and C_4 at time locations. The same ideas are taken to sum the square of errors over all the time positions k to get the equivalent parameters C_3 and C_4 at the frequency dependent representations.

$$\frac{\partial}{\partial c_{3,j}} \sum_{allk} \frac{1}{2^j} \varepsilon_{j,k,y}^2 = 0$$
(5.53)

$$\frac{\partial}{\partial c_{4,j}} \sum_{allk} \frac{1}{2^j} \varepsilon_{j,k,y}^2 = 0$$
(5.54)

From equations (5.51) and (5.52), we obtain the equivalent parameters C_3 and C_4 at the time location presentations:

$$c_{3,k} = \frac{\sum_{allj} \frac{\partial}{\partial k} W_w x (2^j t - k) W_w h_1 (2^j t - k)}{\sum_{allj} \left(\frac{\partial}{\partial k} W_w x (2^j t - k) \right)^2}$$
(5.55)
$$c_{4,k} = \frac{\sum_{allj} W_w y (2^j t - k) W_w h_1 (2^j t - k)}{\sum_{allj} \left(W_w y (2^j t - k) \right)^2}$$
(5.56)

Similarly, from equations (5.53) and (5.54), we obtain the equivalent parameters C_3 and C_4 at the frequency band presentations:

$$c_{3,j} = \frac{\sum_{allk} \frac{\partial}{\partial k} W_w x (2^j t - k) W_w h_1 (2^j t - k)}{\sum_{allk} \left(\frac{\partial}{\partial k} W_w x (2^j t - k) \right)^2}$$
(5.57)
$$c_{4,j} = \frac{\sum_{allk} W_w y (2^j t - k) W_w h_1 (2^j t - k)}{\sum_{allk} (W_w y (2^j t - k))^2}$$
(5.58)

In the case of a stochastic system, the equivalent parameters are obtained by minimizing the expected value of the mean square of the error between the nonlinear system and the linearized system [33], [34].

For
$$\beta_{eq}$$
 and ω_{eq}^2 , one has

$$\frac{\partial}{\partial \beta_{eq,j}} \sum_{allk} E[\varepsilon_{j,k,x}^2] = 0$$
(5.59)

$$\frac{\partial}{\partial \omega_{eq,j}^2} \sum_{allk} E\left[\varepsilon_{j,k,x}^2\right] = 0$$
(5.60)

For c_3 and c_4 , one has

$$\frac{\partial}{\partial c_{3,j}} \sum_{allk} E\left[\varepsilon_{j,k,y}^2\right] = 0 \tag{5.61}$$

$$\frac{\partial}{\partial c_{4,j}} \sum_{allk} E\left[\varepsilon_{j,k,y}^2\right] = 0 \tag{5.62}$$

Using these three methods, we can obtain the responses of the hysteresis system. In the next section, we will show the linearization results by using traditional equivalent and

wavelet-based linearization method.

5.5 Linearization Results

The main objective of this part is to compare the wavelet-based linearization method with the statistical equivalent technique and verify the feasibility of this method. The displacement response of the bilinear hysteresis system under the multi-frequency excitation is compared to numerical solution of the same system response to check the results as well.

We set the system to be a unit mass system with $\beta = 0.02$, $\alpha = 0.5$ and the initial equivalent parameters $\beta_{eq} = 0$, $\omega_{eq}^2 = 0$, the sampling time is 0.01 sec. and time period t = 10.23 sec, $N = 2^{12}$, $x_y = 1$, the hysteretic force: $h(x, \dot{x}, t) = \alpha x + (1 - \alpha)h_0(\dot{x}, y; x_y)$

$$\dot{y} = h_1(\dot{x}, y; x_y)$$

where $h_0 = y \{ u(y + x_y) - u(y - x_y) \} + x_y \{ u(y - x_y) u(\dot{x}) - u(-y - x_y) u(-\dot{x}) \}$ $h_1 = \dot{x} \{ u(y + x_y) - u(y - x_y) + u(y - x_y) u(-\dot{x}) + u(-y - x_y) u(\dot{x}) \}$

The system is excited by the signal $f(t) = 15\sin(4t) + 10\sin(8t) + 5\sin(12t) + 5\sin(16t)$ and the initial conditions are $\beta_{eq} = 0, \omega_{eq}^2 = 0$.

Assume that the system is a linear one to obtain the responses at the different frequencies. By using the procedure demonstrated above and substituting the response to the expressions of equivalent parameters β_{eq} and ω_{eq}^2 in equations (5.42) and (5.43), the parameters converged after 4 times iteration. Therefore, the time varying linear system is obtained and the similar procedure can be applied to the equivalent parameters c_3 and c_4 as well. Thus, the bilinear hysteresis system can be linearzed as this time varying linear system and these results are compared with those responses which are obtained by the statistical method and numerical method. These results are shown in Figures 5.6 and 5.7.



Figure 5. 6 Numerical solution of bilinear hysteretic system to stationary excitation



Figure 5. 7 Wavelet-based linearization method comparing with numerical solution

5.6 Conclusion

In this chapter, a wavelet-based linearization method is developed for bilinear hysteresis system. The equivalent parameters in this bilinear hysteresis systems can be obtained in time dependent and frequency dependent respectively, thus, the nonlinear system is linearized to be the time varying linear system. The simulation results are compared with those of numerical method and the traditional equivalent method to demonstrate this wavelet-based linearization method is a promising tool in linearizing nonlinear systems to
their approximate linear ones.

6 CONCLUSIONS

6.1 Contribution of the Thesis

In this dissertation, a wavelet-based linearization method for approximating the nonlinear system behavior to its equivalent linearized system is presented. This method has been compared with the traditional equivalent technique. The whole procedure for this novel approach has been demonstrated by using the SDOF Duffing oscillator and bilinear hysteresis nonlinear models as examples to verify the feasibility of this method.

The contributions of the thesis are summarized as:

- A comprehensive literature survey on time-frequency analysis and linearization techniques has been carried out.
- The mechanism by which the signal is decomposed and reconstructed using the wavelet transform is presented. The properties and characteristics of some famous harmonic wavelet transforms are investigated.
- A multi-frequency linearization and wavelet-based linearization methods are implemented and compared for Duffing oscillator system when the system is subject to multiple harmonic inputs.
- A wavelet-based linearization method is developed for another nonlinear systembilinear hysteresis system. The wavelet transform for the relationship of input and

output of the linear system is implemented and the differential equation of nonlinear system is expressed in terms of time-frequency localization. The comparison between wavelet-based linearization and statistical equivalent linearization for bilinear hysteresis system has been carried out.

6.2 Future Work

Considering the results of the research project, the author summarizes the following research topics to be investigated in the future:

- It will be worthwhile investigating some other types of nonlinear system to see the feasibilities of wavelet linearization method on different cases. Especially, for some multi-degree freedom system. If the general form of wavelet-based equivalent linearization method can be obtained, it will be a valuable contribution.
- Another possible extension is to find different wavelet basis functions in this approach because so many mother wavelets can be chosen. It might be worthwhile checking some different wavelets for different applications.
- The very promising application is to use this technique to design the nonlinear control system to meet control requirement.

REFERENCES

- Azenha A., Machado, J.A.(1998), "On the Describing Function Method and the Prediction of Limit Cycles in Nonlinear Dynamical Systems." SAME, 33:307-320.
- Bennoune A., Kaddouri and M. Ghribi (2007) "Application of the Dynamic Linearization Technique to the Reduction of the Energy Consumption of Induction Motors." Applied Mathematical Sciences, 1 (34):1685 -1694.
- 3 Agrawal, O.D. (1998). "Application of Wavelets in Modeling Stochastic Dynamic Systems." Journal of Vibration and Acoustics, 120,763-769.
- 4 A. Gelb and W. Van der Welde, (1968) "Multiple Input Describing Functions and Nonlinear System Design." McGraw-Hill Book Company, New York.
- 5 Ahmadi, G. (1980) "Mean Square Response of a Duffing Oscillator to Modulated White Noise Excitation by the Generalized Method of Equivalent Linearization." J. sound and Vibration, 71:9-15.
- 6 Atalik, T.S. and Utku, S. (1975) "Stochastic Linearization of Multi-Degree-Of-Freedom Non-linear Systems," Earthquake Engineering and Structure Dynamics, vol.4, 411-420.
- 7 Asano,K. and W.D Iwan (1998) "An Alternative Approach to the Random

Response of Bilinear Hysteretic Systems." Earthquake Engineering and Structural Dynamics 12:229-236.

- Basu, B., and Gupta, V.K. (1998) "Seismic Response of SDOF Systems by Wavelet Modeling of Non-stationary Processes." Journal of Engineering Mechanics (ASCE), 124(10): 1142-1150.
- Basu, B., and Gupta, V.K. (1999) "On Equivalent Linearization using Wavelet Transform." Journal of Vibration and Acoustics ,121(4): 429-432.
- 10 Basu, B., and Gupta, V.K. (2000) "Stochastic Seismic Response of Single-Degree-of-Freedom-Systems though Wavelet." Engineering of Structural Dynamics, 29:1659-1676.
- Basu, B., and V.K. Gupta, V.K. (2001). "Wavelet Based Stochastic Seismic
 Response of Duffing Oscillator." Journal of Sound and Vibration, 245(2):251-260.
- 12 Booton, R. C., Jr. (1954) "Nonlinear Control Systems with Random Inputs", IRE Trans.Circuit Theory, vol. CT-I, no. 1, pp. 9-17.
- Caughey, T.K. (1960) "Random Excitation of a System with Bilinear Hysteresis."Journal of Applied Mechanics, 27:649-652.
- 14 Caughey, T.K. (1963) "Equivalent Linearization Techniques." Journal of the Acoustical Society of America, 35(11): 1706-1711.

- Caughey, T.K. (1971) "Nonlinear Theory of Random Vibrations." Advanced Applied Mechanics, 11: 209-253.
- 16 Chui, C.K., (1992) "Wavelet Analysis and Its Applications", vol. 1: "An Introduction to Wavelets." Academic Press, Inc., New York.
- 17 Chui, C.K.,(1992). "Wavelet: A Tutorial in Theory and Applications", Academic Press, New York.
- 18 Cox, C.S.(1987) "Algorithms for Limit Cycles Predition: A Tutorial Paper," International Journal for Electrical Engineering Education 24 (2):175-182.
- Daubechies, I. (1988) "Orthogonal Bases of Compactly Supported Wavelets,"Communication on Pure and Applied Mathematics, (41): 909-996
- 20 Daubechies, I., (1990) "The Wavelet Transform, Time-frequency Localization and Signal Analysis." IEEE Transactions on Information Technology 36: 961-1005.
- Daubechies, I. (1992) "Ten Lectures on Wavelets." CBS-NSF Regional Conferences in Application Mathematics.
- 22 Dobson, S., Noori, M., Hou, Z. and Dimentberg, M., (1998) "Direct Implementation of Stochastic Linearization for SDOF Systems with General Hysteresis." Structure Engineering and Mechanics 6(5): 473-484.
- 23 Francisco. L. Silva, Sonia E. Ruiz.Z (2005) "Calibration of Equivalent Linearization Gaussion Approach Applied to Simple Hysteretic System Subjected

to Narrow Band Seismic Motion," Structural Safety 22,211-231.

- Gabor, D. (1946) "Theory of Communication," Journal of IEEE 93(3): 429-457.
- 25 Gelb, A. and Van Der Velde, W.E., (1968) Multiple-input Describing Functions and Nonlinear System Design, McGraw-Hill, New York.
- 26 Grizzle, J.W. and Kokotovic, P.V(1988) "Feedback linearization of Sampled Data Systems," IEEE Trans.Auto.Control.vol.33.857-859.
- 27 Grochenig, K. (2001) Foundations of Time-Frequency Analysis, Birkhauser.
- 28 Grossmann, A. and Morlet, J. (1984) "Decomposition of Hardy Functions into Square Integral Wavelets of Constant Shape." SIAM Journal of Mathematical Analysis 15:723-736.
- 29 Grossmann, A., Kronland-Martinet, R. and Morlet, J. (1989) "Reading and Understanding Continuous Wavelet Transform." In Wavelet Time-Frequency Methods and Phase Space. J.M. Combes, A. Grossmann and Ph. Tchamitchian, eds, pp. 2-20, Berlin: Springer-Verlag.
- Isidori,A.(1989) "Nonlinear Control Systems-An Introduction." Sringer-Verlag.
 New York.
- 31 Iwan, W.D. (1973) "A Generalization of the Concept of Equivalent Linearization."International Journal of Nonlinear Mechanics 8: 279-287.
- 32 Iwan, W.D. (1974) "Application of Nonlinear Analysis Techniques." Appl. Mech.

Earthquake Eng, Appl. Mech. Div., ASME, AMD, 8: 135-161.

- Iwan, W.D., and Mason, A.B., Jr. (1980) "Equivalent Linearization for Systems
 Subjected to Nonstationary Random Excitation." International Journal of
 Nonlinear Mechanics 15: 71-82.
- Iwan, W.D. and Yang, I.M., (1971) "Statistical Linearization for Nonlinear Structures." Journal of Engineering Mechanics ASCE, 97, 1609-1623.
- Iwan, W.D. and Yang, I.M., (1972) "Application of Statistical Linearization to
 Nonlinear Multi-degree-of-freedom Systems." Journal of Applied Mechanics,
 ASME, 39: 545-550.
- 36 Iwan, W. D. and L.D. Lutes (1968) "Response of Bilinear Hysteresis to Stationary Random Excitation." Journal of Acoustical Society of America 43: 545-552.
- 37 Hedrick, J.K. and Girard, A. (2005) "Feedback Linearization Theory and Application." Sringer-Verlag, NewYork. NY.
- 38 Kaise, G. (1994). A Friendly Guide to Wavelets. Boston, MA Birkauser.
- 39 Koichiro, A (1984) "An Alternative Approach to the Random Response of Bilinear Hystertic Systems." Earthquake Engineering and Structural Dynamics.vol.12:229-236.
- 40 Kryov N.N. and Bogoliubov N.N. (1947) "Introduction to Nonlinear Mechaniccs,"

Princeton University, Princeton.

- Mallat, M. (1989) "A Theory for Multiresolution Signal Decomposition: The
 Wavelet Representation." IEEE Pattern Analysis and Machine Intelligence 11(7):
 674-693.
- 42 Mallat, S. (1999) "A Wavelet Tour of Signal Processing.", Academic Press, San Diego, CA.
- Mason, A.B.J. (1997) Some Observations on the Random Response of Linear and Nonlinear Dynamical Systems. Ph. D. Dissertation, Calif. Inst. Of Tech.
 Pasadena.
- 44 Meyer Y.,(1992) "Wavelets and Operations," Cambridge University Press, NewYork.
- 45 Meyer Y.,(1993) "Wavelets Algorithms and its Applications," SIAM. Philadephia PN.
- 46 Newland, D.E. (1993) "An Introduction to Random Vibrations, Spectral & Wavelet Analysis", Longman Scientific & Technical, New York.
- 47 Newland, D.E. (1994) "Harmonic and Musical Wavelets." Proceeding: Mathematical and Physical Sciences, 444(1992):605-620.
- 48 Newland, D.E. (1994) "Wavelet Analysis of Vibration, part 1: Theory." Journal of Vibration and Acoustics 116: 409-416.

- 49 Newland, D.E. (1994) "Wavelet Analysis of Vibration, part 2: Wavelet Maps."Journal of Vibration and Acoustics, 116: 417-425.
- 50 Noori, M.N., Davoodi, H., Choi, J.D.(1986) "Zero and Nonzero Mean Random Vibration Analysis of New General Hysteresis Model." Journal of Probabilistic Engineering Mechanics1, No.4 192-201.
- 51 Papoulis, A. (1962). The Fourier Integral and its Applications. New York, Mc-Graw Hill.
- 52 Patra,K.C. and Singh,Y.P.(1996) "Graphical Method of Prediction ofLimit Cycles for Nonlinear Systems," IEEE Preceeding Control Theory and Applications 143(5) 423-428.
- 53 Rao, R.M. and Bopardikar, A.S., (1998), "Wavelet Transforms: Introduction to Theory and Applications". Addison Wesley Longman Inc.
- 54 Roberts, J.B. and Spanos, P.D. (1990) "Random Vibration and Statistical Linearization", Mineola, New York.
- Ruzzene, M.A. Fasana, et al. (1997) "Natural Frequencies and Damping Identification using Wavelet Transform: Application to Real Data," Mechanical System and Signal Processing 11(2):207-218.
- 56 Sawaragi, Y., Sugai, N., and Sunahara, Y., (1962) "Statistical Studies of Non-linear Control Systems". Nippon Printing and. Publishing Co., Osaka, Japan.

- 57 Sinitsyn, I.N., (1976) "Methods of Statistical Linearization (Survey)." Auto.Remote Control 35: 765-776.
- Spanos, P.D. (1978) "Stochastic Linearization Method for Dynamic Systems with Asymmetric nonlinearities." Report EMERL 1126.
 Engineering Mechanics Research Laboratory, University of Texas at Austin.
- 59 Spanos, P.D. (1981) "Stochastic Linearization in Structural Dynamics," Appl.Mech. Revs., ASME, 34: 1-8.
- 60 Spanos, P.D. and B.A. Zeldin (1978) "On the Existence and Uniqueness of Solution Generated by Equivalent Linearization," International Journal of Nonlinear Mechanics. 13(2): 71-78.
- 61 Sreng G.,(1989) "Wavelet and Dilation Equations: A Brief Introduction." SIAM.Review.vol.31.No.4 614-628.
- 62 Steven Chingyei Chuang, Ph.D.,P.E. and Jen-So Huang, Ph.D. "Describing Functions For Effective Stiffness and Effective Damping of Hystersis Structures," Electronic Journal of Structural Engineering. 4 (2004).
- Stoker, J.J (1965) "Nonlinear Vibrations in Mechanical and Electrical Systems",
 Interscience Publishers, New York.
- 64 Yuan, J. and Wang,Y. (1999) "Describing Function Method for Steady-State Analysis of practical Sliding Model Control Systems." IEEE Transactions on

Automatic Control. vol.AC-22(2):212-222.

- 65 Ville, J., (1948) "Theory of Applications on the Notion of Signal Analysis,"Cables Transmission. vol.2:61-74.
- 66 Walker J., (1999) "A Primer on Wavelets and their Scientific Applications."Chapman&Hall CRC, NewYork.
- 67 Wen, Y.K, (1980) "Equivalent Linearization of Hysteretic System Under Random Excitation." Journal of Application Mechanics. ASME 150-154.
- 68 Wigner, E.P.,(1939) "On the Quantum Correction for Thermodynamic Equilibrium." Physical Review, vol.40: 749-759.
- 69 Misiti M., Misiti Y., Oppenheim G., Poggi J. –M, (1996), "Wavelet Toolbox for Use with Matlab", The Matheworks Inc., Natick MA.
- Allen J.B. and Rabiner,(1997), "A unified approach to a short-time Fourier analysis and synthesis,"IEEE Proceedings.vol.65(11), 1558-1564
- 71 Burrus C.S.,Gopinath R.A and Guo H.(1998) "Introduction to wavelets and wavelet transforms: a primer,"Prentice Hall.Upper Saddle River NJ
- 72 Carmona R., Hwang W.J., Torresani B., (1998) "Practical time-frequency analysis," Academic Press, NewYork
- 73 Kobayash M.,(1998) "Wavelets and their applications," SIAM, Philadelphia, PA
- 74 Kovacevic J. and Cohen A.,(1996) "Wavelets:the mathematical

background,"Proceedings of the IEEE,vol.84,no.4 514—522

- 75 Vetterli M. and Kovacevic J.,(1995) "Wavelets and sub-band coding," Prentice Hall,Englewood Cliff NJ
- Wickerhauser M.V.,(1994) "Adaptive wavelet analysis: from theory to software,"AK Peters
- Wang W.J. and McFadden P.D.,(1995) "Application of orthogonal wavelets to early gear damage detection," Mechanical Systems and Signal processing.9(5) 497-507